THE EFFECT OF THE CREDIT CRUNCH ON OUTPUT PRICE DYNAMICS: THE CORPORATE INVENTORY AND LIQUIDITY MANAGEMENT CHANNEL*

RYAN KIM

I study how a credit crunch affects output price dynamics. I build a unique micro-level data set that combines scanner-level prices and quantities with producer information, including the producer's banking relationships, inventory, and cash holdings. I exploit the Lehman Brothers failure as a quasi-experiment and find that the firms facing a negative credit supply shock decrease their output prices approximately 15% more than their unaffected counterparts. I hypothesize that such firms reduce prices to liquidate inventory and generate additional cash flow from the product market. I find strong empirical support for this hypothesis: (i) the firms that face a negative bank shock temporarily decrease their prices and inventory and increase their market share and cash holdings relative to their counterparts, and (ii) this effect is stronger for the firms and sectors with a high initial inventory or small initial cash holdings. *JEL* Codes: E31, E32, E44, G01, G31, G33, L11.

*This article is a revised version of the main chapter of my Ph.D. dissertation at Columbia University. I am indebted to Emi Nakamura, Jón Steinsson, Jonathan Vogel, and David Weinstein for invaluable discussions, guidance, and support and to Robert Barro, Pol Antràs, and five anonymous referees for excellent editorial advice and suggestions. I am grateful to Hassan Afrouzi, Hie Joo Ahn, JaeBin Ahn, George Alessandria (discussant), Henrique Basso (discussant), Jeffrey Campbell, Chris Conlon, Olivier Darmouni, Andres Drenik, Simone Gilchrist, Matthieu Gomez, Katherine Ho, Colin Hottman, Jay Hyun, Yuko Imura (discussant), Yang Jiao, Yoon Joo Jo (discussant), Réka Juhász, Lorena Keller (discussant), Amit Khandelwal, Seok Ki Kim (discussant), Pravin Krishna, Jennifer La'O, Mauricio Larrain, Jae Won Lee (discussant), Christian Moser, Nathalie Moyen (discussant), Andreas Mueller, Hyunseung Oh, Matthias Paustian, Ricardo Reis, Giacomo Saibene (discussant), Bernard Salanié, José Scheinkman, Stephanie Schmitt-Grohé, Mine Senses, Samer Shousha, Jae Sim, Joseph Stiglitz, Philip Strahan (discussant), Catherine Thomas, Martin Uribe, Eric Verhoogen, Daniel Villar, Daniel Wolfenzon, Michael Woodford, Egon Zakrajšek, and participants in numerous seminars and conferences for helpful comments. Part of this article was written at the Federal Reserve Board while I was a dissertation intern; I am thankful for their support. I thank Dave Harvey, Sue Hutchinson, and Jing Zhou for their generous help with the data collection and analysis. All errors are my own.

[©] The Author(s) 2020. Published by Oxford University Press on behalf of the President and Fellows of Harvard College. All rights reserved. For Permissions, please email: journals.permissions@oup.com

The Quarterly Journal of Economics (2021), 563–619. doi:10.1093/qje/qjaa025. Advance Access publication on July 8, 2020.

I. INTRODUCTION

The questions of how and the extent to which credit market disruptions affect firms' economic decisions have been of vital interest in the economics and finance literature, particularly after the 2007–2009 financial crisis. This period was characterized not only by a significant drop in total output and employment but also by a dysfunctional credit market. At the peak of the credit market stress following the September 2008 failure of Lehman Brothers, new loans to large borrowers dropped 79% relative to the level of new loans to large borrowers in the credit boom period (Ivashina and Scharfstein 2010). The TED spread, an indicator of perceived credit risk, surpassed 300 basis points after the Lehman Brothers failure, which broke the previous record set after the 1987 Black Monday crash.

During these credit market disruptions, the producer price index plummeted approximately 15% in three months.¹ Given this aggregate correlation, this article seeks to answer the following questions. Do the firms that face a negative credit supply shock decrease their output prices, and if so, why?

Identification poses the biggest challenge in answering these questions. Although there was a clear positive correlation between inflation and credit market conditions during the financial crisis, it is difficult to identify the true relationship between these two factors from the aggregate data. The aggregate correlation is based on the Great Recession, and without relying on credit market conditions, conventional macroeconomic models can easily explain a decrease in inflation during the recession. Even worse, many influential events, such as a fall in housing prices (Mian, Rao, and Sufi 2013), a drop in oil prices (Hamilton 2009), and a decrease in international trade (Eaton et al. 2016), occurred at the same time, which makes aggregate time-series comparisons nearly impossible.

To overcome this identification challenge, I build a novel micro-level data set that combines the following: producers' prices and sales at the barcode level from the Nielsen Homescan Panel database; producers' balance sheet information from the Orbis database; and producers' loan market access from the Dealscan

^{1.} See Online Appendix S3. This price movement partially reflects a fall in oil and commodity prices in this period, and I discuss the relationship between the change in commodity prices and the mechanism proposed in this article in explaining the aggregate inflation dynamics.

database. The merged data set contains detailed information on prices and quantities sold by public and private firms and the producers' banking relationships from 2004 to 2011. For example, if a household purchases Coke at a store, then I observe the price and quantity of the Coke purchased, Coca-Cola's balance sheet, and which bank Coca-Cola deals with. To the best of my knowledge, this article is the first to combine information on producers' price and quantity with information on their banking relationships.

With detailed micro-level data, I exploit the "bank shock" at the time of the Lehman Brothers failure and find that the firms that face a negative credit supply shock decrease their output prices approximately 15% more than their unaffected counterparts. Although these micro-level data provide rich cross-sectional variation in addition to time-series variation, they do not automatically solve the identification problem because of the difficulty in identifying credit-constrained firms in the data. Farre-Mensa and Ljungqvist (2016) test conventional micro-level financial constraint measures such as Kaplan-Zingales (Kaplan and Zingales 1997) and Whited-Wu (Whited and Wu 2006) and conclude that they do not accurately identify financially constrained firms because they are constructed using firm-level balance sheet variables that likely reflect company characteristics other than their level of financial constraint. Thus, instead of relying on firm-level balance sheet variables, to generate plausibly exogenous variation in firm-level credit supply conditions I use a change in bank health at the time of the Lehman failure. In addition to my main measure of the change in bank health based on banks' loan issuance. I use the following three bank shock measures from Chodorow-Reich (2014) that are not highly correlated but provide consistent results: banks' exposure to the Lehman failure; banks' exposure to toxic asset-backed securities; and bank balance sheet items, such as bank deposits and net trading revenues that are unlikely to be correlated with borrowers' characteristics. These three measures affect firms' credit supply conditions for reasons that are plausibly orthogonal to their characteristics related to pricing decisions.

I hypothesize that the firms that face a negative credit supply shock decrease their output prices by liquidating inventory and dumping their products to generate extra cash flow from the product market, and I provide strong empirical support for this hypothesis. By using micro-level data and a corresponding identification strategy, I find that the firms that face negative bank shocks decrease their inventory relative to their counterparts. These firms decrease their output prices only temporarily and then increase them after approximately one year; these actions indicate that firms temporarily liquidate their inventory because of a negative credit supply shock but cannot sell their inventory forever; thus, they must increase prices in the medium run. In addition, these firms increase their market share and cash holdings. which illustrates that they increase their cash flow by selling more to the product market as a result of lowering their output prices. Moreover, the effect on output prices is stronger for the firms or sectors that had larger inventories or smaller cash holdings before the Lehman failure, which confirms my hypothesis. From a corporate inventory and liquidity management perspective, this hypothesis can be interpreted to imply that companies convert illiquid assets (inventory) to liquid assets (cash) when their insurers (banks) cannot lend to them and that companies decrease their output prices in this conversion process.

I also estimate heterogeneous treatment effects across firms and sectors and implement numerous robustness tests to gain additional insights from the data and to confirm the validity of the bank shock measures. I find that the firms facing negative bank shocks decrease their output prices more if (i) they face high product demand elasticity, (ii) they did not issue a bond before the credit supply shock was realized, (iii) they had to pay out loans immediately after the Lehman failure, (iv) they dealt with a small number of lead lenders in the pre-Lehman period, or (v) they are small in terms of employment or total assets. The firms that face high demand elasticity are more likely to decrease their output prices when they face a negative credit supply shock because they can sell more products while experiencing a smaller decrease in output prices.² If demand elasticity is very low—such that products complement other products—then firms would not be able to cut output prices to increase revenue. Other results are also intuitive and consistent with the literature since the effect of a credit supply shock is likely to be larger for the firms that do not have bond access (Becker and Ivashina 2014), that had to

2. Of course, the firms that face high demand elasticity might have a smaller incentive to lower their prices if their goal is to generate a certain amount of revenue. Such companies would be able to make enough revenue by decreasing their prices by a small amount, whereas their counterparts must lower their prices more to earn the same amount of revenue. There are two opposing forces, and the empirical question of which effects dominate remains.

pay out loans after the Lehman failure (Almeida et al. 2012), or that are small (Gertler and Gilchrist 1994). Moreover, I undertake various additional empirical analyses to address the potential concerns that relate to retailer decisions, purchaser characteristics, variety-quality changes, external validity, changes in local conditions, foreign exposure, other price indices, pretrends, and sample weights.

My findings are surprising because they seemingly conflict with the influential work of Gilchrist et al. (2017), who, by using liquidity as a measure of financial constraint, find that financially constrained firms raise their output prices. The underlying reason for this difference is the difference in the measure of financial constraint, which is the "weak liquidity position" in Gilchrist et al. (2017). The phrase "weak liquidity position" is used in their paper and refers to firms with a small amount of liquidity. I replicate their findings in my sample by using their measure of financial constraint—liquidity—to confirm that different results arise from the difference in the measure, not the sample or regression specification. Consistent with Gilchrist et al. (2017), my results are robust to using two alternative periods of the liquidity positions, namely, the contemporaneous period (2008) and the lagged period (2006).

A natural question is why different measures of financial constraint cause different results. Previous studies in the corporate finance literature raise concerns about using liquidity as a measure of financial constraint. In their study on liquidity position, Kahle and Stulz (2013) find that the firms that face a negative bank shock raised—rather than lowered—their liquidity in 2008.³ This result is consistent with my findings in Table V and with my hypothesis that such firms convert inventory (illiguid assets) to cash (liquid assets). The firms that suffer from a negative bank shock would therefore be classified as firms in a "strong liquidity position," not a "weak liquidity position." Regarding the lagged (2006) liquidity position, a seminal paper by Bates, Kahle, and Stulz (2009) identifies more than 10 factors that lead firms to hold more liquid assets. In particular, they find that the "weak liquidity position" is associated with more investment, borrowing, acquisitions, and a stable cash flow-characteristics that likely reflect unconstrained companies rather than constrained

^{3.} See, for example, Perez-Orive (2016) and Garcia-Macia and Villacorta (2019) for a theoretical formulation of such behavior.

companies. I confirm the findings of Bates, Kahle, and Stulz (2009) by using 2006's liquidity position and find that, controlling for such characteristics, the effect of lagged liquidity on firm-level price changes signs and becomes insignificant. More generally, a vast body of literature in corporate finance asks why companies hold liquidity. Almeida et al. (2014) survey this literature and conclude that firms hold more liquidity because they are more likely to be financially constrained. This argument dates back to Keynes (1936), who discusses that there is a fundamental relationship between corporate liquidity management and financial friction and emphasizes the precautionary saving motive to explain the variation in the corporate liquidity position. Because of these concerns about using liquidity as a measure of financial constraint, I instead use bank shocks-which are not subject to this criticism—as proposed by Ivashina and Scharfstein (2010) and Chodorow-Reich (2014). See Section V and Online Appendix S7 for a more detailed discussion and empirical evidence on the reconciliation with the previous study.

More broadly, this article is related to the publications that study financially constrained companies' pricing decisions. Standard business cycle models with financial friction emphasize the cost-push channel—in which an increase in output prices is due to an increase in financial costs-or other channels that lead companies to increase their prices because of financial friction.⁴ However, this increase in output price will be inconsistent with the micro-level empirical evidence in this article. I seek to expand these previous studies by incorporating a fire sale inventory mechanism, which maintains consistency with the micro-level empirical evidence that I find and is also consistent with a sudden, dramatic, and temporary fall in inflation in this period; at the end of the analysis, I present the back-of-the-envelope calculation for the effect of the credit supply shock on aggregate inflation dynamics. Papers on industrial organization and corporate finance also study this topic, but they are inconclusive regarding how financial distress affects output prices, particularly at the aggregate level. Several papers on the airline industry find that financial distress leads to a decrease in output prices (Borenstein and Rose 1995;

4. Papers such as Del Negro, Giannoni, and Schorfheide (2015), Christiano, Eichenbaum, and Trabandt (2015), and Gilchrist et al. (2017) incorporate financial friction into a business cycle model to explain inflation dynamics during the Great Recession.

Phillips and Sertsios 2013), whereas other studies find the opposite result for retail industries (Chevalier 1995a, 1995b; Chevalier and Scharfstein 1995, 1996). I complement this line of research by exploiting a new data set with bank shocks that generate plausibly exogenous variations in companies' credit supply conditions.

This article emphasizes the importance of inventory and liquidity management in explaining the output price dynamics during the banking crisis and is closely related to previous studies on this topic. The most closely related papers to this study are the seminal work by Gertler and Gilchrist (1994) and Kashyap, Lamont, and Stein (1994), who provide evidence that liquidityconstrained firms liquidate their inventories.⁵ At the international level, Alessandria, Kaboski, and Midrigan (2010b) document a close link among inventory, the price-cost markup, and the interest rate shock. In business cycle research, inventory is known to contain valuable information because of its volatility and procyclical behavior (Ramey and West 1999).⁶ In extending previous studies, I show that the bank shock, which has rarely been addressed in this literature, can generate procyclical inflation and inventory dynamics.

For the empirical analysis, I use the bank shock to overcome the identification challenge. Previous studies document that the firms that cannot borrow from banks are likely to default (Khwaja and Mian 2008) and decrease their investment (Peek and Rosengren 1997, 2000; Amiti and Weinstein 2018), employment (Chodorow-Reich 2014; Greenstone, Mas, and Nguyen 2020), and exports (Amiti and Weinstein 2011; Paravisini et al. 2015). I show that such a shock has an essential effect on output prices through the inventory and cash adjustment of firms.

5. Other papers, such as Carpenter, Fazzari, and Petersen (1994, 1998) and Bates, Kahle, and Stulz (2009), also suggest a close link between corporate inventory investment and internal finance (or the corporate cash position). My findings also complement the studies that examine how firms substitute between external financing and internal financing or between banks and cash (e.g., Sufi 2009; Lins, Servaes, and Tufano 2010; Campello et al. 2011).

6. Previous studies examine inventory dynamics and the sources of cyclical fluctuation (West 1990), the slope of marginal cost (Ramey 1991), price-cost markup cyclicality (Bils and Kahn 2000; Kryvtsov and Midrigan 2013), international trade (Alessandria, Kaboski, and Midrigan 2010a, 2011), international business cycles (Alessandria, Kaboski, and Midrigan 2013), and news shocks (Crouzet and Oh 2016). Papers such as Fisher and Hornstein (2000) and Khan and Thomas (2007) incorporate inventory into the business cycle model to explain the salient feature of the data. The rest of the article is structured as follows. Section II explains the construction and description of the micro-level data, credit supply shock, and price index. Section III presents the main reduced-form empirical results. Section IV proposes the inventory adjustment hypothesis with empirical support, and Section V clarifies the relationship with Gilchrist et al. (2017). Section VI presents a streamlined discussion of a simple business cycle model, which considers aggregate dynamics. Section VII concludes.

II. DATA DESCRIPTION AND MEASUREMENT OF THE VARIABLES

II.A. Data Description

A major novelty of this article is that it constructs a microlevel data set that integrates producers' output prices and quantities, their inventories and cash holdings, and their relationships with banks.

The price and quantity data originate from the ACNielsen Homescan Panel, which was made available by the Kilts Marketing Data Center at the University of Chicago Booth School of Business.⁷ The data contain approximately 1.7 million barcodelevel product prices and quantities recorded daily from 55,000 households a year on average. A barcode is a unique universal product code (UPC) assigned to each product and is used to scan and store product information. The data begin in 2004 and end in 2011, which covers the period before, during, and after the financial panic of 2008. All households sampled by Nielsen are provided with in-home scanners to record their purchases of products with barcodes. Nielsen assigns a sample weight—or a projection factor—to each household based on 10 demographic variables to make the sample nationally representative.⁸ According to Nielsen, the Homescan Panel covers approximately 30% of all household

7. Researcher(s) own analyses calculated (or derived) based in part on data from the Nielsen Company (US) and marketing databases provided through the Nielsen Datasets at the Kilts Center for Marketing Data Center at the University of Chicago Booth School of Business. The conclusions drawn from the Nielsen data are those of the researcher(s) and do not reflect the views of Nielsen. Nielsen is not responsible for, had no role in, and was not involved in analyzing and preparing the results reported herein.

8. The 10 demographic variables are household size, household income, head of household age, race, Hispanic origin, male head education, female head education, head of household occupation, the presence of children, and the Nielsen county size.

expenditures on goods in the consumer price index (CPI) basket. To confirm the validity of the data, Online Appendix S3.A shows that a scanner price index made from the Nielsen data closely follows the U.S. Bureau of Labor Statistics (BLS) official price index.

There are many advantages of using the ACNielsen database to identify the effect of credit supply shocks on output price dynamics. First, the database records product prices at the barcode level, which is likely to be the most granular way to define the product. This feature not only helps uncover the effects of the introduction and destruction of products on prices but also allows a comparison of similar products produced by the firms that face different degrees of a credit supply shock. Second, the data set provides product sales information, which is useful for separating the quality component of product prices and for confirming that the effect is not driven by the change in product demand. Finally, the data record the detailed characteristics of purchasers, such as income and employment, the location and retail store where products were purchased, and product-level information such as product unit and size. This information is valuable for addressing other potential identification concerns related to the change in the purchasers' income and employment, the housing price, local conditions, and the retailers' behavior.

I integrate the prices and quantities of each product with its producer by using the GS1 U.S. Data Hub. GS1 is the company that issues barcodes to producers.⁹ Their data record the company name and address for each barcode-level product, which provides a way to link barcode-level product information with its producer information. A "firm" in the database is defined based on the entity that purchased the barcodes from GS1. Thus, a firm in the data could be a manufacturer, such as a Coca-Cola manufacturer that needs a barcode to sell its cherry-flavored 500-ml Diet Coke, or it could be a small retailer that wants to sell private-label products. In the final sample, manufacturers account for approximately 62% of total sales, retailers account for approximately 33% of total sales, and other entities, such as wholesalers, account for less than 5% of total sales before the Lehman Brothers failure.

^{9.} GS1 provides a business with up to 10 barcodes for a \$250 initial membership fee and a \$50 annual fee. There are significant discounts on the cost per barcode for the firms that purchase larger quantities of barcodes (see http://www.gs1us.org/get-started/im-new-to-gs1-us).

To collect producer information. I further combine the ACNielsen-GS1 matched database with the Orbis and Fixed Income Securities Database (FISD). Orbis is the firm-level data set compiled by Bureau van Dijk (BvD) and has detailed administrative, financial, production, and ownership information for both public and private firms. The data set records the firms' inventory and cash holdings, which are particularly helpful in testing the "fire sale" of inventory hypothesis. It also includes information such as the detailed four-digit NAICS industry codes, the number of foreign subsidiaries and branches, total assets, and the number of employees, which allow me to conduct additional empirical analvses and robustness checks. Similar to the Nielsen data set, the data cover the period from 2004 to 2011; the data set was downloaded from the BvD proprietary online browser for Orbis data.¹⁰ The online platform of the Orbis database provides software that automatically matches firms based on their name, address, industry code, and other information available in both the Orbis data and other data. I exploit this feature to merge the GS1 data and the corresponding barcode-level information with all other firmlevel and bank-level information, including the FISD.¹¹ The FISD records historical corporate bond issuance and ratings and is used to extract information on the producers' bond market access.

Finally, to extract information on bank lending to each producer, I include the Dealscan database. The Dealscan database contains comprehensive historical information on loan pricing and contract details, terms, and conditions. It mainly includes information on the syndicated loan market, in which more than one bank arranges a loan to a firm. The process usually begins with one or more lead arrangers signing a preliminary loan contract called a "mandate," and these arrangers retain part of the loans and raise the rest of the funds from the participants. For each loan (or facility/package), the data include information on its purpose (e.g., corporate purposes or debt repayment), type (e.g., term

10. The Orbis data used in the main analysis were downloaded in 2014. Downloading at this time maximizes the number of years that can be used alongside the Nielsen data set because of how BvD manages the Orbis database. First, only the most recent 10 years of the sample are available on the online platform. If I had downloaded data in 2015, then I would have missed the firm-level information for 2004. Second, there is a reporting lag of two years in the database (Kalemli-Ozcan et al. 2015). If I had downloaded data in 2013, then the coverage of 2011 (and 2012) would likely be incomplete.

11. I also hand-checked the validity of the merged sample.

loan or revolving line of credit), amount, interest spread, maturity, and lender information, which identifies the lead arranger and the lender's contribution to each loan. In constructing the credit supply shock, I used loans identified as serving a corporate purpose or serving as working capital. Carey and Hrycay (1999) show that the data record between one-half and three-fourths of the volume of outstanding commercial and industrial (C&I) loans in the United States, and Chava and Roberts (2008) discuss that the coverage of Dealscan data increased from 1995 onward.¹²

I supplement the combined data with Zillow housing price data and Current Population Survey (CPS) data on homeownership to specifically address the drop in housing prices and homeownership during this period. In addition, I merged several bank-level variables used by Chodorow-Reich (2014) that reflect a change in bank health at the time of Lehman's failure, demand elasticities from Hottman, Redding, and Weinstein (2016), industry-level inventory information from the NBER-CES database (Bartelsman, Becker, and Gray 2000), and an industrylevel external financial dependence index (Rajan and Zingales 1998) measured by using firm-level cash and expenditure information from the Compustat database.

Table I reports the summary statistics of the combined sample. The merged data set includes approximately 200 firms identified from the Orbis firm classification (BvD identification number) that were active in the syndicated loan market and that sold their products. See Online Appendix S4 for further details about the data used in this article. I dropped all firms that entered or exited after the Lehman failure to abstract away from firm dynamics. The median firm in the sample sells 30 products (UPCs) in approximately three product groups, such as pet food and school supplies. These 200 firms are relatively large compared to other firms in the consumer packaged goods market, where most firms are extremely small.¹³ Although this discrepancy raises concerns

12. If I use the 2004–2011 sample, then the share of loans covered in Dealscan relative to the total C&I loans is approximately 77%. This number should be interpreted with caution due to the discrepancy between the two databases, such as the unit of measure (e.g. flow versus stock) and coverage (e.g. regulated banks versus unregulated banks). See Ivashina and Scharfstein (2010) for a more detailed discussion of the coverage of the Dealscan data during the financial crisis.

13. My sample considers approximately one-fifth of the sales and one-fourth of the total number of purchases in the Nielsen data. Originally, there were slightly under 15,000 firms in the Orbis database integrated with the ACNielsen Homescan

LE I	ROTHERS FAILURE PERIOD (2006Q
TAI	STATISTICS FOR THE PRE-LEHMAN

Variable	N	Mean	Std. dev.	$_{ m p10}$	p50	p90
Panel A: Firm-group variables						
$ ilde{P}_{fg}$	2,055	3.90	5.92	1.21	2.47	6.80
Sales (millions \$)	2,055	28.51	110.34	0.04	1.36	56.08
Market share (%)	2,055	4.84	12.59	0.01	0.37	14.40
Average # of UPCs per quarter	2,055	31.43	76.01	1	10	72.33
Average # of buyers per quarter (millions)	2,055	2.33	9.11	0.00	0.12	4.76
Panel B: Firm variables						
ΔL_f	200	0.47	0.18	0.26	0.45	0.69
Lehman exposure	198	0.84	0.36	0.50	0.74	1.28
ABX exposure	198	1.06	0.28	0.81	1.01	1.34
Bank items	198	44.90	12.99	28.17	46.63	58.46
Bond issuance (binary)	200	0.28	0.45	0	0	1
Listed status (binary)	200	0.36	0.48	0	0	1
Firm age	198	47.82	35.87	13	35	97
Median spread (bp)	187	150.77	106.34	25	150	300
Average maturity (month)	197	53.65	15.21	32.5	60.0	61.0
Number of groups	200	10.28	19.28	1	က	26
Panel C: Group variables						
Demand elasticities across UPCs	100	8.13	4.25	5.02	6.93	14.06
Demand elasticities across firms	100	4.45	2.04	2.62	3.92	7.33
Number of firms	100	20.55	7.74	10.5	20.5	31.0

purpose was classified as corporate use or as working capital. An ure animular, accurate to the main measure of bank shock constructed from the change in firm-group-specific price index, is defined in Section II.C. Market share is the sales share of firm *f* in group *g*. ΔL_f is the main measure of bank shock constructed from the change in firm-group-specific price index, is defined in Section II.C. Market share is the sales share of firm *f* in group *g*. ΔL_f is the main measure of bank shock constructed from the change in firm-group-specific price index, is defined in Section II.C. Market share is the sale share of the *f* in group *g*. ΔL_f is the main measure of bank shock constructed from the change in firm-group-specific price index, is defined in Section II.C. Market share is the sale share of the *f* in group *g*. ΔL_f is the main measure of bank shock constructed from the change in firm-group-specific price index, is defined in Section II.C. Market share is the sale share of the *f* in group *g*. ΔL_f is the main measure of bank shock constructed from the change in the loan in the loan intervent of the sale of the sale of the sale share of the loan intervent of the sale of the sale of the sale of the loan intervent of the loan intervent of the loan intervent of the sale of the sale of the sale of the loan intervent of the sale loans issued by the bank, as described in Section II.B. Lehman exposure is the percentage of the bank's syndication portfolio in which Lehman Brothers had a lead role in the loan deal. The ABX exposure variable equals the loading of the bank's stock return on the ABX AAA 2006-H1 index between October 2007 and December 2007. The bank items variable is the sum of bank deposits and net trading revenue divided by total assets. All three measures are defined and discussed in Section II.B. The demand elasticities are defined based on the nested CES demand system in Hottman, Redding, and Weinstein (2016), as discussed in the Appendix.

574

THE QUARTERLY JOURNAL OF ECONOMICS

about the representativeness of the sample, the effect is likely to be at most underestimated, given that small firms are more sensitive than large firms concerning a credit supply shock. In addition, there remains a large degree of heterogeneity across firms and groups in the sample. The largest firm-group pair sells 130 times more UPCs than the median firm-group pair in the sample, and only approximately one-third of the firms in the sample are publicly listed or issued bonds before the Lehman failure. I exploit this variation to confirm that the effect of the credit supply shocks is larger for small firms. In addition, I confirm my findings by using various sample weights in the regression analysis and by conducting an external validity check with more representative data.

A potential concern of using the combined database is that the information on prices and quantities is collected from households, whereas the credit supply shock, inventory, and liquidity are measured at the firm level. Thus, the manufacturers' prices and sales information in the data might reflect the decision of retailers (or wholesalers) that purchase products from manufacturers and sell to households instead of the decision of the manufacturers that face the negative supply shock. To extract adequate firm-level prices and quantities from the data, I follow Hottman, Redding, and Weinstein (2016) and aggregate the variables across retailers within firms. Given that a large fraction of manufacturers deal with a large number of retailers in the data, the idiosyncratic behaviors of retailers are likely to disappear for many manufacturers at the firm level.¹⁴ In addition, even if there are retailers (or wholesalers) that are large enough to affect the aggregated firm-level manufacturer prices, according to the previous papers that study incomplete pass-through (e.g., Burstein and Gopinath 2014), such an effect would at most underestimate the main finding. I discuss this point in detail and conduct three additional empirical analyses, such as collecting and using the prices measured at the manufacturer level and using only retailers in the

Panel, and most of these firms are dropped when I require that firms in the sample be active in the syndicated loan market before and after the Lehman failure. Most of these dropped samples do not have valid firm-level information, such as employment or total assets, in the Orbis data. It is likely that Orbis could not record the balance-sheet information for these exceptionally small firms. See Online Appendix S4.B.

^{14.} There are 187 retailers per firm on average during the pre-Lehman period in the final sample.

sample, which confirms the robustness of the main findings in Online Appendix S6.B.

II.B. Credit Supply Shock

I follow Chodorow-Reich (2014) to construct ΔL_f , a credit supply shock measure, which simply and coherently extracts the information on changes in the firms' access to credit as a consequence of a change in bank health.

I choose two periods, pre- and post-Lehman, to measure the credit supply shock to exploit the Lehman failure, which is known to be surprising and dramatic. The post-Lehman period takes up the three quarters immediately after the Lehman failure: 2008:Q4 to 2009:Q2. During this time, the TED spread, the measure of perceived credit risk, increased dramatically (Figure I, Panel A). At the same time, the number of loans and loan amounts issued plummeted, and the interest spread spiked (Figure I, Panels B and C). The pre-Lehman period corresponds to the same three quarters in the earlier years, at the time of the credit market expansion: 2005:Q4 to 2006:Q2 and 2006:Q4 to 2007:Q2. These quarters were chosen to minimize seasonality concerns. To compare the extreme periods, I did not use the period immediately before the Lehman failure (2007:Q4 to 2008:Q2) for my main regression analysis.¹⁵ However, additionally defining this period as a pre-Lehman period does not alter the result, as shown in Online Appendix S6.A. In fact, given the modest degree of financial market stress, this period is useful for a placebo test of the validity of the measure of Lehman exposure. In Online Appendix S6.A, I show that the Lehman shock did not affect prices during this period.

Based on this timing, I construct the measure of bank shock as follows. Given the change in bank health measures, I take a weighted average of bank health for each firm to generate the firm-specific credit supply shock:

(1)
$$\Delta L_f = \sum_{b \in S_f} \alpha_{fb, \text{last}} \Delta(\text{Bank Health})_{-f, b},$$

15. Total commercial and industrial loans also did not fall in this period. However, this is due to an increase in credit drawdowns by corporate borrowers on existing credit lines, not the issuance of new loans (Ivashina and Scharfstein 2010).



(B) NUMBER OF LOANS AND INTEREST SPREAD

(C) AMOUNT OF LOANS AND INTEREST SPREAD

FIGURE I

Timing of the Credit Supply Shock (ΔL_f)

Panel A plots the TED spread, Panel B plots the total number of loans and the average interest spread, and Panel C plots the total amount of loans and the average interest spread. The pre-Lehman period includes the following six quarters: 2005:Q4–2006:Q2 and 2006:Q4–2007:Q2. The post-Lehman period includes the following three quarters: 2008:Q4–2009:Q2. The TED spread, which measures the perceived credit risk, is defined as the difference between the three-month T-bill and the interbank borrowing rate. The number (amount) of loans is the total number (amount) of loans issued according to the Dealscan database, and the interest spread is the amount that the borrower pays in basis points over LIBOR for each dollar drawn down and is averaged across loans within each quarter. The Lehman failure occurred in September 2008 at the end of 2008:Q3.

where $\Delta(\text{BankHealth})_{-f, b}$ is a measure of the firm-bank-specific change in bank health defined in equation (2), and weight $\alpha_{fb, \text{last}}$ is bank *b*'s share of the total amount of the last syndicated loan it made to firm *f* before the Lehman failure.¹⁶ S_f denotes the set of banks that lend to firm *f* for the last syndicated loan that firm

16. The weight reflects the fact that multiple banks arrange a loan to a firm and that different banks lend different amounts for a particular loan. The Dealscan database reports only approximately one-third of α_{jbt} among total loans. I impute the missing α_{ibt} by using the method in Chodorow-Reich (2014).

Downloaded from https://academic.oup.com/gje/article/136/1/563/5868795 by Aalto University Library user on 30 October 2023

f borrowed before the Lehman failure. For example, consider the J.M. Smucker Company, which is famous for its fruit spreads and peanut butter. Suppose that it borrowed from two banks—Chase Bank and Citibank—for its last loans before the Lehman failure and that 80% of its loans were borrowed from Chase Bank and 20% were borrowed from Citibank. I used 0.8 and 0.2 as the weights to take a weighted average of the changes in bank health for Chase Bank and Citibank, respectively, to measure the credit supply shock faced by Smucker's. Although I used the last loan share as a weight to maximize the effect of the bank shock on firms, using the average loan share of the whole pre-Lehman period does not alter the results, as shown in Online Appendix S6.F. This finding is likely attributable to the stability of firm-bank relationships.

The $\Delta(\text{BankHealth})_{-f, b}$ is given by the following expression:

(2)

$$\Delta(\text{Bank Health})_{-f,b} = \frac{\sum_{j \neq f} \alpha_{jb,\text{post}} \times \mathbb{1}(b \text{ lent to } j \text{ post-Lehman})}{\frac{1}{2} \sum_{j \neq f} \alpha_{jb,\text{pre}} \times \mathbb{1}(b \text{ lent to } j \text{ pre-Lehman})}$$

where 1() is an indicator variable equal to 1 if what is in parentheses is true and is 0 otherwise; α_{jbt} denotes bank *b*'s share of the total amount of the loan for each syndicated loan that it made to firm *j* in period *t*. I divide the denominator by 2 to balance the periods as the pre-Lehman period consists of six quarters, whereas the post-Lehman period consists of three quarters.

Roughly, equation (2) is a change in the number of loans issued by banks: the number of loans made by bank b in the post-Lehman period over the number of loans made by bank b in the pre-Lehman period. There are two additional complications. First, to reflect the importance of each loan issued by bank b. I multiply the weight α_{ibt} for each loan made by bank *b* to firm *j*. Second, I intentionally omit firm f from the summation to generate the firm-f-bank-b-specific change in bank health. This "leaveone-out" method partially eases concerns related to the credit demand channel. For example, consider again the example of Chase Bank, which lends to Smucker's and other companies. If I use Chase Bank's loan to Smucker's to measure Chase's change in bank health, then this measure might reflect the change in credit demand that arises from Smucker's product market decisions or financing policies rather than from the change in Chase's willingness to supply credit to Smucker's. To address this concern,

in constructing Smucker's credit supply shock, I examine Chase's lending to all firms excluding Smucker's, for both the pre- and post-Lehman periods. I do the same to measure the change in Citibank's bank health and then take a weighted average across Chase and Citibank to construct Smucker's credit supply shock, as shown in equation (1).¹⁷

To assess the validity of the credit supply shock measure. I check the sample balance and find no significant difference in firm characteristics across the credit supply shock. I first regress the pre-Lehman firm-level characteristics on the credit supply shock that I constructed. As shown in Table II, the credit supply shock is not correlated with the purchasers' characteristics¹⁸ or with the firms' access to the loan market, listed status, bond market access, age, size, or loan characteristics. These results suggest that the measure of credit supply shock constructed for this period reflects the change in bank health rather than borrower or purchaser characteristics. In addition, I implement a test introduced in Khwaja and Mian (2008) and conducted in Chodorow-Reich (2014) to check for selection on unobserved firm characteristics in my sample. Consistent with Chodorow-Reich (2014). I find that the unobserved firm characteristics are balanced. The details of this analysis are reported in Online Appendix S6.I.

In addition to the measure of the credit supply shock constructed above, to confirm the findings I use three bank-level measures of the change in bank health as instrumental variables. The three measures are (i) the banks' exposure to Lehman, (ii) the banks' exposure to asset-backed securities (ABX), and (iii) bank statement items that are unlikely to be correlated with borrower characteristics. Lehman exposure is the fraction of a bank's

17. Note that I use the number of loans instead of loan amounts. I do this to minimize the measurement error due to the imputation of α_{fbt} . However, using loan amounts does not change the results, as reported in Online Appendix S6.F. These results are likely because the drop in loans during this period is driven largely by a change in the number of loans rather than by the loan amounts. The loan size remained stable in this period (Darmouni forthcoming).

18. Housing prices are available at the ZIP code level from the Zillow database, homeownership is available at the county level from the census, and income, employment, education, and household size are available from Homescan Panel data. For each measure, I first take a weighted average across households (or ZIP code or county) within firms and groups by using the sample weight; then I take a sales-weighted average across groups within each firm to construct the measures reported here.

	(1)	(2)	(3)	(4)	(2)	(9)
	ln(Housing price)	Home ownership	ln(Income)	Employment	Education	Household size
Unit		(%)		(\mathscr{Y}_{0})	(years)	(number)
$\Delta L_{ m f}$	-0.03	0.00	-0.00	0.00	-0.03	0.02
	(0.02)	(0.00)	(0.02)	(0.01)	(0.05)	(0.03)
R^2	0.01	0.01	0.00	0.00	0.00	0.00
obs	202	202	202	202	202	202
	Number of loans	Amount of loans	Bond	List	Age	Multi lead
Unit		(\$b)	(D)	(D)	(years)	(D)
$\Delta L_{ m f}$	0.93	2.12	-0.08	-0.06	2.07	-0.02
	(1.16)	(2.33)	(0.06)	(0.07)	(5.53)	(0.04)
R^2	0.00	0.00	0.01	0.00	0.00	0.00
obs	206	206	206	206	204	206
	Spread (median)	Maturity	Total assets	$\operatorname{Employment}$	Inventory/asset	Cash/asset
Unit	(\mathbf{pp})	(month)	(\$m)	(\mathbf{k})	(26)	(%)
$\Delta L_{ m f}$	-14.95	1.61	7.14	73.00	-0.00	-0.02
	(23.87)	(2.40)	(7.49)	(63.16)	(0.03)	(0.02)
R^2	0.00	0.00	0.03	0.03	0.00	0.01
obs	191	203	121	109	72	73
Note. *p ·	/ 10 *** / 05 *** / 01·+ho	strandoud and amond protocol	In the Province of the Provinc	titi - the strength of the str	· · · · · · · · · · · · · · · · · · ·	

It is 1,1000, and bp is basis points. Inforoising preci, none ovversing in nuncourse, employment, curreation, and none-now date accentees a use and employment are firm-fired works in the number and employment are firm-level variables in Orbis and are averaged across 2004, 2005, and 2006. The number and the amount of loans are total sums, multi lead and maturity are averages, and the spread is a median across loans within the pre-Lehman period. Bond access is equal to 1 if the companies issue bonds in 2004;63 to 2007; Q2 and is 0 otherwise.

Downloaded from https://academic.oup.com/qje/article/136/1/563/5868795 by Aalto University Library user on 30 October 2023

580

TABLE II

THE QUARTERLY JOURNAL OF ECONOMICS

syndication portfolio in which Lehman Brothers had a lead role. This measure relies on the notion that certain banks dealt more with Lehman Brothers than others did and decreased their lending relatively more than other banks decreased their lending after the Lehman collapse. According to Ivashina and Scharfstein (2010), this pattern occurs because the borrowers that had a credit line in which Lehman Brothers had a lead role aggressively drew down as a precautionary motive when the lead lender became bankrupt, which drained the liquidity of other lenders that dealt closely with Lehman. The bank's exposure to asset-backed securities is the correlation between its daily stock return and the return on the ABX AAA 2006-H1 index. This index generates variation in the changes in bank health due to the banks' exposure to the toxic residential mortgage-backed securities issued during the second half of 2005. Finally, the bank statement items variable is the sum of a bank's net trading revenue—where many subprime write-downs occurred—and bank deposits divided by its assets before the Lehman Brothers failure. For reasons that are plausibly orthogonal to a borrower's pricing decision, all three measures are likely to generate variation in a change in bank health.¹⁹ For each bank-specific change in a bank health measure, I construct a firm-level credit supply shock by following equation (1). The correlations among these three variables are weak at the firm level in my sample, which generates a presumably independent variation in the producer's credit supply condition.²⁰

II.C. Firm-Group Price Index

This article adopts the nested CES utility function in Hottman, Redding, and Weinstein (2016) to build the firmgroup-specific price index from the ACNielsen Homescan Panel database.²¹ The utility-based price index has two main advantages over other indices; the index is consistent with the standard models that use the CES utility function, including the model discussed in Section VI, and it incorporates the effect of product

19. I am grateful to Gabriel Chodorow-Reich for making these measures available on his website.

20. Corr(Lehman, ABX) = 0.04, Corr(ABX, BankItem) = 0.06, Corr(Lehman, BankItem) = 0.44.

^{21.} This framework is isomorphic to the nested logit demand system in which heterogeneous consumers demand a single product in each stage (Anderson, de Palma, and Thisse 1992) and is more flexible than the standard CES utility function (Atkeson and Burstein 2008; Edmond, Midrigan, and Xu 2015).

quality and variety changes on output prices. In particular, the index can be exactly decomposed into the conventional price index and the quality-variety correction and allows me to quantify the effect of the credit supply shock on prices through quality and variety changes. Although previous studies generally identified the importance of such quality and variety changes in explaining output price dynamics (Nakamura and Steinsson 2012; Hottman, Redding, and Weinstein 2016), the effect of the credit supply shock on the quality-variety margin of the output price index is negligible as shown in Online Appendix S2.A. This result is largely consistent with the hypothesis proposed in this article. If the firms that face a negative credit supply shock liquidate inventory and decrease price to generate extra revenues from the product market, then it is not obvious why they would change their product quality or variety to adjust their output prices.²²

Given that the quality-variety margin does not make much difference in the main results, for readability, I present the simple conventional price index that subtracts the quality-variety correction from the utility-based price index. The Appendix presents the nested-CES demand system with the derivation of the utilitybased price index. Although all the tables in the main body of the article use conventional price indices, the results are robust to using the utility-based price index. Using other conventional price indices, such as the Laspeyres, Paasche, and the Törnqvist price index, does not change the results, as shown in Online Appendix S6.G.

The conventional price index is a simple geometric average of prices across UPCs within the firm, product group, and time (quarter):

(3)
$$\tilde{P}_{fgt} = \left(\prod_{u \in \Omega_{fgt}} P_{ut}\right)^{\frac{1}{N_{fgt}}},$$

where subscript u is the UPC or barcode-level product, f is the firm, g is the product group, and t is time. Ω_{fgt} is the set of the UPCs made by firm f in product group g at time t, and N_{fgt} is the number of UPCs made by firm f in product group g at time t. P_{ut} is the price of the UPC at time t and is measured by dividing the total sales by

^{22.} However, the credit supply shock influences the market share through the quality adjustment. See Online Appendix S2.B.

quantities at the UPC-time-level. Note that the aggregate price index measured from this geometric average price index follows the official BLS price index closely, as shown in Online Appendix S3.A. Adding the quality-variety correction term to this geometric average price index recovers the utility-based price index.

III. THE EFFECT OF THE CREDIT CRUNCH ON THE OUTPUT PRICE

In three complementary ways, this section presents the effect of a credit supply shock on producers' output price dynamics. First, I visualize the effect by plotting the aggregate price indices of two groups of firms that face different degrees of credit supply shocks. Second, I confirm the visualization by conducting more rigorous regression analyses with the continuous measure of a credit supply shock, three other instruments, and a rich set of control variables. Finally, I draw an event study plot to show the dynamic effect of the credit supply shock.

To visualize the effect, I divide my sample into two categories based on the measure of the credit supply shock (ΔL_f) defined in equation (1). One category of firms faces a negative credit supply shock larger than the 80th percentile of the shock's size distribution, and the other category of firms faces a negative credit supply shock smaller than the 20th percentile of its distribution.²³

For each category of firms, I measure the price index by taking a geometric average of the firm-group-time-level price index (\tilde{P}_{fgt}) defined in equation (3) across firms within product group and time:

(4)
$$\tilde{P}_{gt,c} = \left(\prod_{f \in \Omega_{gt,c}} \tilde{P}_{fgt}\right)^{\frac{1}{N_{gt,c}}},$$

where $\Omega_{gt,c}$ is the set of firms in product group g and category c at time t, and $N_{gt,c}$ is the number of firms in product group g and category c at time t. Similar to the firm-group-level price index, this product group-level price index is the part of the nested-CES utility-based price index that does not adjust for a varietyquality correction. I aggregate this index across product groups

^{23.} Setting different thresholds, such as the 75th and the 25th percentile of the distribution, yields similar results.



FIGURE II

Differential Change in Price Indices

The differential change in the price index between credit-constrained firms and their unaffected counterparts. The red dashed line (color version available online) denotes the quarter-level price index of the firms that face a large negative credit supply shock, and the blue dotted line denotes the quarter-level price index of the firms that face a small negative credit supply shock. The vertical solid red line shows the timing of the Lehman failure, which is used to measure the credit supply shock, as shown in Section II.B.

within category and time by using the following Törnqvist price index:

(5)
$$\frac{\tilde{P}_{t,c}}{\tilde{P}_{t_0,c}} = \prod_{g \in \Omega_{t,c}} \left(\frac{\tilde{P}_{gt,c}}{\tilde{P}_{gt_0,c}}\right)^{\frac{\varphi_{gt,c}+\varphi_{gt_0,c}}{2}}$$

where $\Omega_{t,c}$ is the set of the product groups g in category c at time t. t_0 is the base time (2004:Q1) and $\varphi_{gt,c}$ is a market share of product group g at time t in each category. The same procedure is used to compare the scanner price index made from Nielsen data with the BLS official price index in Online Appendix S3.A.

Figure II plots the price index measured in equation (5). Although the category-specific measure of the price index does not fully use the credit supply shock variation across firms, the figure clearly illustrates the main empirical results in this article. The two price indices, which are made by using two different categories of firms, follow each other closely before the shock. This empirical pattern reflects that pricing behaviors are similar across the firms that face different degrees of a credit supply shock in the preshock period. However after the shock, the price index of the firms facing a larger negative credit supply shock decreases relative to the price index of their counterparts. The gap between these two indices persists for several quarters after the shock but eventually closes at the end, which shows that the effect is temporary. In addition, this figure makes it clear that the micro-level empirical analyses in this article study the relative changes in prices and are not vulnerable to aggregate shocks, such as the aggregate demand changes in this period.

To confirm the visualization, I more rigorously examine the effect of a credit supply shock on producers' output price dynamics by using the following specification:

(6)
$$\Delta \ln \tilde{P}_{fg} = \lambda_g + \beta (-\Delta L_f) + \theta X_f + \varepsilon_{fg},$$

where subscript f is the firm and g is the product group. \tilde{P}_{fg} is the firm-group-specific price index that I constructed from the AC-Nielsen barcode-level data discussed in Section II.C. In measuring \tilde{P}_{fg} , to make the price index comparable to the credit supply shock, I take the geometric average of \tilde{P}_{fgt} in equation (3) across quarters within 2006:Q4–2007:Q2 (the last three quarters in the pre-Lehman period) and 2008:Q4–2009:Q2 (the post-Lehman period).

Then I take the difference of the logged price index across the pre- and post-Lehman periods. ΔL_f measures the change in the firm-level credit supply as a result of the deterioration of bank health, as discussed in Section II.B. I change the sign of ΔL_f to interpret β as a result of a negative credit supply shock on output prices. X_f includes the initial and lagged firm-level control variables. λ_g is allowed in the regression to compare product prices within product groups. I weighted the regression by the initial total sales in each product group and firm to reveal the aggregate dynamics, similar to the analysis in Amiti and Weinstein (2018). Using different regression coefficient, does not change the results, as shown in Online Appendix S6.E. β is the coefficient of interest

that measures the effect of a credit supply shock on the change in output prices.

The key identification assumption to make a causal interpretation of β is that any confounding factors that affect a firm's pricing decisions do not simultaneously affect its lender's lending to other firms. Concerning this assumption, the biggest identification threat is that the demand shock can potentially affect both the firms' pricing decisions and their previous lenders' lending decisions toward other borrowers. For example, the large drop in housing prices in this period affected different consumers differentially (Mian, Rao, and Sufi 2013). Thus, across the products made by different firms, these consumers may purchase products differentially, and in turn these firms could differentially demand from their lenders a different number of loans. If these firms are large enough for the lender to cut its lending to other borrowers, then my assumption is violated.

I argue that the key assumption in this article is well supported. The narrative evidence suggests that bank health deterioration in this period originated from the Lehman Brothers failure (Ivashina and Scharfstein 2010), real estate and toxic assets (Santos 2011), and the bank liability structure (Fahlenbrach, Prilmeier, and Stulz 2012) rather than from the corporate loan sector. The fact that the corporate loan sector did not cause the credit market disruption in this period is particularly true for the consumer packaged goods market used in the sample; relative to consumers in other sectors, these consumers did not change their purchasing behavior dramatically. The empirical pattern of the aggregate price and quantity of loans during this period supports this view. After the Lehman bankruptcy, there was not only a dramatic drop in the number and amount of loans but also a sudden large increase in the interest spread (Figure I, Panel B and C). This credit market behavior suggests that there was a shift in credit supply rather than in credit demand, at least at the aggregate level.

Additionally, I allow a rich set of initial and lagged firm-level characteristics (X_f) in this regression to address potential spurious correlations. To control for firms' liquidity substitution from loan markets to bond markets when banks cannot provide a loan (Becker and Ivashina 2014), I include a pre-Lehman bond rating and the bond issuance information for each firm. The fixed effects of the firms' four-digit NAICS industry, listed status, and a firm size indicator are included to compare firms within these

categories. To address the differential degree of loan market access for each firm. I control for the number and amount of loans that firms received in the pre-Lehman period and for the number of loans that matured in the post-Lehman period because firms would suffer more if they had to pay out their loans in the post-Lehman period (Almeida et al. 2012). Furthermore, to make a reliable comparison across firms. I control for the firm age, the type of the last loan (term loan versus revolver/line), the year the last loan was issued, whether a firm dealt with multiple lead banks, and the last loan's interest spread and maturity. I also add a lagged change in the output price index to control for the potential pretrend. In addition, Nielsen data provide detailed purchaser characteristics, such as income, education, employment, age, and household size. I merge the data with housing price data at the ZIP code level from the Zillow data and homeownership at the county level from the census data. Adding these purchasers' characteristics does not change the results, as shown in Online Appendix S6.C. Note that the observed pre-Lehman borrower and purchaser characteristics are balanced, as shown in Table II.

To confirm my findings. I use the following three instruments that are not highly correlated but generate plausibly exogenous variations in the firms' credit supply conditions: Lehman exposure, ABX securities exposure, and bank statement items. These measures are used as instrumental variables to consistently interpret the coefficients. Using the cross-sectional variation in Lehman exposure is a well-established identification strategy used in the literature to study bank, fund, and firm behavior (e.g., Ivashina and Scharfstein 2010; Aragon and Strahan 2012; Chodorow-Reich 2014; Darmouni forthcoming). Before its failure, Lehman Brothers was the fourth-largest investment bank and had more than \$600 billion in assets, and its collapse was surprising and dramatic. By using this instrument, I effectively assume that what happened to companies such as Smucker's in the consumer packaged goods market did not lead Lehman to bankruptcy. This assumption is very persuasive, given the ample evidence that the Lehman failure was due to the bank's risky lending, investment strategy, and toxic mortgage-backed securities holdings. Using ABX exposure or bank statement items (the sum of bank deposits and net trading revenue to assets) also generates a credit supply variation that is plausibly uncorrelated with the factors affecting companies' pricing decisions. The firststage regression results for the three instruments are reported in

		$\Delta L_{ m f}$			
	(1)	(2)	(3)		
Lehman exposure	-0.359***				
	(0.088)				
ABX exposure		-0.262^{***}			
-		(0.093)			
Bank items			0.422^{***}		
			(0.108)		
Firm-level controls	Yes	Yes	Yes		
Product group FE	Yes	Yes	Yes		
$E[\Delta L: IV_{p90} - IV_{p10}]$	-0.247	-0.361	0.479		
Observations	1,658	1,658	1,658		

TABLE III	
FIRST-STAGE REGRESSION	J

Notes. *p < .10, **p < .05, ***p < .01; the standard errors are clustered by firm and product group, and the regression is weighted by initial sales. The firm-level controls are the firm's listed status, four-digit NAICS fixed effects, age, size, bond rating, number of loans, amount of loans, loan type, loan-year fixed effects, multi-lead fixed effects, number of loans due in the post-Lehman period fixed effects, loan spread, loan maturity, and lagged $\Delta \ln P_{fg}$.

Table III. All the estimated coefficients are intuitive and statistically significant.

In Online Appendix S6, I also conduct numerous robustness checks regarding concerns such as product quality and variety, retailers' decisions, local conditions, purchaser behavior, foreign exposure, initial cash holdings, pretrends, and external validity. In particular, I use the quantity information available in the Nielsen data and find that the firms that face a negative credit supply shock increase their market share. If the negative credit supply shock measures the negative demand shock, the firms facing this shock would experience a decrease in market share rather than an increase in market share. I show this result and rationalize it with the inventory adjustment hypothesis in Section IV.A.

The other assumption of the regression analysis is the longterm firm-bank relationship or the existence of switching costs when companies must form new relationships with banks. If companies can quickly change to other banks when their previous lenders cannot issue loans, these companies might not be affected by a bank shock. However, it is very unlikely that firms can easily find a new lender quickly because of the adverse selection for switchers that prevents lenders from providing new loans. In addition, the monitoring cost is likely to decline more for repeated borrowers, which also eases the moral hazard problem for lenders. This relationship lending is especially true for the United States, where the Secretary of the Treasury has made KYC (Know Your Customer) mandatory for all U.S. banks since 2002. As a result of this regulation, there is a nontrivial implicit cost for U.S. banks in establishing new relationships with customers. In particular, I examine a credit market disruption period, during which banks are especially hesitant to form new relationships.²⁴

Table IV shows the empirical results based on equation (6). The first column reports the OLS regression results without conditioning on the control variables. The coefficient is negative and statistically significant, which confirms what Figure II presents with more variation in the measure of the credit supply shock. The second column adds all the firm-level control variables, including the product group and NAICS four-digit industry code fixed effects. Although adding the firm-level controls increases the estimated coefficient slightly, adding the fixed effects raises the effect substantially, which emphasizes the importance of comparing products in the same groups and comparing the firms that sell the same primary product industry code. Columns (3)-(6) present the IV regression results obtained by using Lehman exposure. ABX securities exposure, bank statement items, and all three variables together as instrumental variables. Regardless of whether OLS is used with the main credit supply shock variable or which instruments are used, the estimated coefficients are negative, statistically significant at the 5% level, and quantitatively similar.

The regression results clearly show that the firms that face a negative credit supply shock decrease their output prices significantly relative to the output prices of their counterparts. I standardize the credit supply shock measure (ΔL_f) to interpret the coefficient. Conditioned on the control variables, a one standard deviation increase in the negative credit supply shock decreases output prices by approximately 8%. If I compare extremely credit-constrained firms and credit-unconstrained firms in the sample by looking at the 90th:10th percentile ratio, the effect is approximately 15 to 18 percentage points. In Online Appendix S6.J, I confirm the empirical results by showing no pretrends with the same regression specification and credit supply shock but with a

^{24.} Empirically, Chodorow-Reich (2014) confirms this sticky firm–bank relationship with a regression analysis.

	Δ				
		$\ln \tilde{P}_{ m fg}$: 2006q4–2007	7q2 to 2008q4–200	9q2	
	OLS		$(-\Delta L_f) ext{ instr}$	umented using	
(1)	(2)	Lehman (3)	ABX (4)	BankItem (5)	All (6)
$(-\Delta L_f)$ -2.31***	-8.31^{***}	-7.13^{**}	-7.36^{**}	-7.31^{**}	-7.25^{***}
(0.52)	(1.85)	(3.13)	(3.03)	(3.42)	(1.97)
Firm-level controls No	Yes	Yes	Yes	Yes	Yes
Product group FE No	Yes	Yes	Yes	Yes	Yes
Four-digit NAICS FE No	Yes	Yes	Yes	Yes	Yes
First-stage <i>F</i> -statistic		16.70	7.90	15.20	11.90
J-statistic p -value					1.00
$E[\Delta \ln \tilde{P}]$ 11.4	11.4	11.4	11.4	11.4	11.4
$E[\Delta \ln \tilde{P}: \Delta L_{p90}-\Delta L_{p10}]$ -5	-18.1	-15.6	-16.1	-15.9	-15.8
Observations 1,658	1,658	1,658	1,658	1,658	1,658

TARLE IV

listed statús, age, bond rating, number of loans, amount of loans, loan type, loan-year fixed effects, multi-lead fixed effects, number of loans, due in the post-Lehman period fixed effects, loan spread, loan maturity, and lagged $\Delta \ln P_{fg}$.

590

THE QUARTERLY JOURNAL OF ECONOMICS

change in the log price index in previous periods, from 2004:Q4–2005:Q2 to 2006:Q4–2007:Q2.

In addition to the main regression analysis, I conduct an event study analysis based on the measure of Lehman exposure by using the following regression specification:

(7) $\ln P_{fg,t} - \ln P_{fg,t-4} = \lambda_{g,t} + \beta_t (-\Delta L_f) + \theta X_{f,t} + \varepsilon_{fg,t},$

where t is the quarter, not the pre- and post-Lehman periods. For this firm-quarter-level analysis, I use the full utility-based price index defined in the Appendix to minimize the measurement errors associated with product entry and exit. Rather than instrumenting the main measure, the measure of Lehman exposure is directly used in the regression as a reduced form.²⁵ Based on this regression analysis, I estimate the effect of a credit supply shock for all quarters in the data.

The estimated coefficients are plotted in Figure III. The figure reveals the transparent dynamic effect of the bank shock on output price dynamics. The estimated coefficients are not statistically different from zero before the Lehman failure, which suggests that there is no pretrend. However, at the time of the Lehman failure, the coefficients are negative for the first two quarters and near zero for the subsequent quarters, which shows that the firms that faced a negative credit supply shock decreased their output prices. After a year, however, the estimated coefficients become positive for approximately three quarters and return to zero for the remaining quarters. This plot clarifies that the effect is temporary and that firms increase their output prices in the medium and long run. Note that Figure II and Figure III show that the effect is temporary, but the duration of the effect is shorter in the latter figure. This is largely because the regression analysis used in Figure III allows more variation in the credit supply shock and has additional control variables, including fixed effects.

The fire sale inventory hypothesis is fully consistent with this temporary effect. If it is true that firms decrease their output prices by liquidating inventory and dumping their products on the market, then they would not be able to sell their inventories forever, and thus they must accumulate inventory at some point,

^{25.} Instrumenting the main credit supply shock with the Lehman exposure does not change the qualitative results.



FIGURE III

Price Dynamics and Credit Market Disruptions

This figure is based on equation (7), and the measure of the Lehman exposure is directly used as a reduced form. The parameter estimates $(\hat{\beta}_t)$ for each quarter are plotted. A 95% confidence interval is reported for each estimated coefficient, standard errors are clustered by firm and product group, and the regression is weighted by initial sales. The firm-level controls are the firm's listed status, fourdigit NAICS fixed effects, age, bond rating, the number of loans, the amount of loans, the loan type, loan-year fixed effects, multi-lead fixed effects, the number of loans due in the post-Lehman period fixed effects, loan spread, and loan maturity.

which suggests that the effect should be temporary. I discuss this hypothesis in detail in the next section.

IV. MECHANISM: FIRE SALE OF INVENTORY

IV.A. Inventory, Market Share, Liquidity, and Employment

The result in the previous section seems to be counterintuitive, as most studies interpret financial distress as an increase in credit cost and therefore predict an increase in output prices due to a negative credit supply shock.²⁶

^{26.} The publications that emphasize the effect of financial cost on output price include Barth and Ramey (2002), Del Negro, Giannoni, and Schorfheide (2015), and Christiano, Eichenbaum, and Trabandt (2015). Other mechanisms are discussed in the literature. For example, Gilchrist et al. (2017) places more emphasis on



FIGURE IV

Aggregate Inventory and Corporate Cash Holdings

This figure plots the total business inventory and corporate cash holdings. The aggregate inventory data are downloaded from the FRED Economic Data. The aggregate corporate cash holding is measured by using the quarterly Compustat database, which was downloaded from the WRDS. It is aggregated across firms within quarters after excluding financial firms and utilities and is seasonally adjusted by using the X-13ARIMA-SEATS Seasonal Adjustment Program from the Census. More information on the Compustat data is given in Online Appendix S4.

I propose a hypothesis that can rationalize the empirical finding. I call it the fire sale of inventory hypothesis. When firms face a negative credit supply shock and cannot borrow from banks, they have an incentive to aggressively liquidate their inventories and sell their products at a low price to generate extra cash flow from the product market. From a corporate liquidity management perspective, the firms that cannot borrow from their lenders try to accumulate cash by selling off their inventory at low prices to generate extra cash flow.

Figure IV plots the total business inventory and corporate cash holdings to see whether this hypothesis is plausible. The

consumer habits, and Chevalier and Scharfstein (1996) emphasize both consumer habits and strategic interaction in explaining firms' price-setting behavior due to financial friction.

aggregate inventory increased before the financial panic, but in the middle of the financial panic, it plummeted.²⁷ Aggregate corporate cash holding seems to be stable and does not correlate with the aggregate inventory before the financial panic, but amid the financial panic, it rises dramatically. Note that the period of financial panic when the total inventory fell and corporate cash holdings rose is precisely the time when the aggregate output price indices decreased temporarily, as shown in Online Appendix S3. These movements in aggregate business inventory, corporate cash holdings, and output price indices suggest that the proposed hypothesis is plausible, at least at the aggregate level.

To rigorously support the hypothesis, I return to the microlevel data and the corresponding identification strategy. Identical to equation (6), except for the dependent variable, the following regression specification is used:

(8)
$$\Delta Y_{fg} = \lambda_g + \gamma (-\Delta L_f) + \theta X_f + \varepsilon_{fg},$$

where ΔY_{fg} equals four different dependent variables, namely, a change in inventory, market share, cash holding, and employment.²⁸ Note that the market share is the only firm-group specific variable among the four dependent variables and that productgroup fixed effects are not allowed in the regression for other variables. ΔL_f is the credit supply shock constructed in Section II.B, and X_f is the vector of corresponding firm-level control variables.

I provide strong empirical support for the fire sale of inventory hypothesis, as shown in Table V.²⁹ Consistent with the hypothesis, I observe inventory holding at the firm level and find that the firms that face a negative credit supply shock liquidate

27. Using other measures of aggregate inventory, such as the real manufacturing and trade inventories or the aggregated measure of inventory from the Orbis database, shows a similar pattern.

^{29.} I report only the regression results instrumented with the measure of Lehman exposure in Table V, but the qualitative results are consistent if I use other credit supply shock measures.

Y_{fg}	Inventory _f (1)	Market Share _{fg} (2)	Cash Holding _f (3)	Employment, (4)
$(-\Delta L_f)$ instrumented	-30.1^{**}	2.4^{**}	5.6^{***}	-23.5^{**}
using Lehman	(13.4)	(1.2)	(1.8)	(10.9)
Firm-level controls	Yes	Yes	Yes	Yes
Product group FE	No	Yes	No	No
First-stage F-statistic	32.7	17.8	67.1	26.5
$E[\Delta \ln Y: (-\Delta L_{p90}) - (-\Delta L_{p10})]$	-51.8	5.25	11.3	-38.2
Observations	992	1,658	1,286	1,453

TABLE V

FIRE SALE OF INVENTORY HYPOTHESIS: EMPIRICAL SUPPORT

Notes. p < .10, p < .05, p < .05, p < .01. For the firm-level regressions in columns (1), (3), and (4), the standard errors are clustered by the three-digit NAICS, the regression is weighted by initial Y_f , and the firm-level controls are a firm's listed status, two-digit NAICS fixed effects, number of loans, multi-lead fixed effects, loan spread, number of loans due in the post-Lehman period fixed effects, and bond rating. For the firm-group-level regression in column (2), the cluster groups of standard errors, regression weights, and control variables are identical to the specification used in Table IV.

their inventories.³⁰ Such firms increase their market share, which suggests that they generate extra sales from the product market by selling off their inventory. Such firms accumulate more cash, which implies that they convert inventories (illiquid assets) to cash (liquid assets). These firms lay off workers, an action that is a well-known result in the literature.

Table V clearly shows the importance of inventory in generating an output price fall due to the adverse credit supply shock. If one thinks of employment as a proxy for production, then the firms that face a negative credit supply shock decrease their production based on column (4).³¹ Without inventories, such firms that reduce their production would not have enough products to supply the market and would probably increase their output prices at the equilibrium. This reaction of firms is a conventional shift in the supply curve effect that leads to a rise in output prices. However, with inventories, such firms still increase their market share or sales (column (2)), because they draw down their inventories (column (1)) to provide an additional supply of products and accumulate cash (column (3)) from the product market. Inventory plays

30. In Online Appendix S5, I analyze with Compustat data the part of the inventory that is affected by the credit supply shock. The effect on inventory mainly comes from the final-good inventory and raw materials inventory.

31. Measuring production as sales minus inventory and regressing this measure on the credit supply shock similarly shows that the firms that face a negative credit supply shock decrease their production.

the role of a wedge or gap between the sales and production and makes production and sales move in opposite directions, which thus changes the direction of output price movements. Note that the increase in market share is small, despite a large decrease in prices and inventory. In Online Appendix S2.B, I present supporting evidence that a fall in firm-level product quality because of the negative credit supply shock lowers the market share, which partially counteracts the positive effect of the price changes on the market share.

In addition, I use the initial inventory and liquidity position across firms and industries together with the credit supply shock measure and find that the results also support the inventory adjustment hypothesis. The next section discusses this analysis in detail.

IV.B. Heterogeneous Treatment Effect

I exploit the rich firm heterogeneity and group heterogeneity in the sample and estimate a heterogeneous treatment effect to provide additional insights and confirm the empirical findings in the previous section. I use the following regression specification:

(9)

$$\Delta \ln \tilde{P}_{fg} = \lambda_g + \beta_1 (-\Delta L_f) \times Z_{fg} + \beta_2 (-\Delta L_f) + \beta_3 Z_{fg} + \theta X_f + \varepsilon_{fg},$$

where Z_{fg} represents the firm-level or firm-group-level characteristics, such as inventory or cash holdings, before 2007. The only difference between this specification and equation (6) is the presence of Z_{fg} , which allows the effect of a credit supply shock on output prices to vary across the firm and group characteristics. The major assumption that I make in this regression is that firms or industries do not anticipate the sudden drop in their previous lenders' bank health after the Lehman failure and thus do not endogenously hold more Z_{fg} to hedge against this particular credit supply shock. Given that the Lehman failure in 2008 was surprising and that the signs of the mortgage crisis became apparent in 2007, this assumption is plausible. Note that the coefficients β_2 should be interpreted with caution because, with the exception of indicator variables, Z_{fg} is always positive.

First, I provide additional support for the fire sale of inventory hypothesis, as shown in Table VI, Panel A. According to the hypothesis, firms that had a large amount of inventory before

	$\Delta \ln \tilde{P}_{\rm fg}$: 2006q4–2007q2 to 2008q4–2009q2				
	(1)	(2)	(3)	(4)	
Panel A: Inventory an	d liquidity				
Z_f	Inventory	Ind. inventory	RZ index	Loan due	
$(-\Delta L_{\rm f}) \times Z_{\rm f}$	-5.26^{***}	-16.98^{***}	-7.50^{***}	-11.38^{**}	
	(0.90)	(5.17)	(2.29)	(5.69)	
$(-\Delta L_{\rm f})$	-0.67	10.10	-11.89^{***}	-5.67^{***}	
	(11.91)	(53.86)	(4.08)	(2.12)	
Firm-level controls	Yes	Yes	Yes	Yes	
Product group FE	Yes	Yes	Yes	Yes	
Observations	808	496	496	1,797	
Panel B: Alternative f	nancing and	size			
Z_f	Bond access	# of lead lenders	Total assets	Employment	
$(-\Delta L_{\rm f}) \times Z_{\rm f}$	5.83^{**}	1.98**	9.00***	6.52^{***}	
	(2.73)	(0.99)	(1.46)	(0.92)	
$(-\Delta L_{\rm f})$	-5.91^{***}	-10.63^{**}	-137.33^{***}	-63.02^{***}	
	(2.23)	(4.33)	(22.69)	(10.22)	
Firm-level controls	Yes	Yes	Yes	Yes	
Product group FE	Yes	Yes	Yes	Yes	
Observations	1,800	1,800	834	834	
Panel C: Demand elas	ticity				
$Z_{f\sigma}$	Elasticity	y w/ Bertrand	Elasticity	w/ Cournot	
$(-\Delta L_{\rm f}) \times Z_{\rm for}$	-1.62^{**}	-1.64^{**}	-2.08***	-2.24^{**}	
8	(0.69)	(0.74)	(0.77)	(0.86)	
$(-\Delta L_{\rm f})$	1.46		2.17		
	(2.26)		(2.36)		
Firm-level controls	Yes	No	Yes	No	
Firm FE	No	Yes	No	Yes	
Product group FE	Yes	Yes	Yes	Yes	
Observations	1,800	1,764	1,800	1,764	

TABLE VI

HETEROGENEOUS TREATMENT EFFECTS

Notes. *p < .10, **p < .05, ***p < .01; the standard errors are clustered by firm and product group, and the regression is weighted by initial sales. The firm-level controls are the firm's listed status, four-digit NAICS fixed effects, bond rating, number of loans, amount of loans, loan type, loan-year fixed effects, lean spread, loan maturity, firm age, and pre-Lehman market share. For columns (1) and (2) of Panel A, $(-\Delta L_f)$ interacted with firm-group-level 2006 sales and the NAICS four-digit value of shipment are additionally controlled for, respectively. Inventory is firm-level 2006 log inventory, ind. inventory is NAICS four-digit 2001–2006 average log inventory, and the RZ index is the NAICS four-digit external financial dependence index as in Rajan and Zingales (1998). Loan due is a dummy variable that equals 1 for the firms that borrowed these loans, which matured in the post-Lehman period, before the post-Lehman period. Bond access is a dummy variable for the firms that issued bonds before the post-Lehman period, and # of lead lenders denotes the number of lead lenders for the last pre-Lehman loan. Elasticity is the demand elasticity under the nested CES demand system and different market structures (Bertrand and Cournot), as in the Appendix and Hottman, Redding, and Weinstein (2016).

the Lehman failure should drop their prices more aggressively than firms with small initial inventory because the firms with a large amount of inventory have more inventory to sell. I test this prediction by using the variation in the 2006 inventory holdings conditional on initial sales.³² The effect of the negative credit supply shock is stronger by approximately 5.3% if firms hold an additional 1% of inventory stock, consistent with the hypothesis. The key assumption that firms did not store inventory in preparation for the credit crunch is supported in Table II and consistent with the model in Section VI, where producers hold inventories to avoid the stock-out of products. Given that this assumption is more plausible at the industry level, I also use industry-level initial inventory and confirm this finding in Table VI, Panel A, column (2).

The inventory adjustment hypothesis also predicts that to ensure cash flow, the firms that are in urgent need of cash should drop their prices more aggressively than the firms with a large amount of liquidity. I find that when they face a negative bank shock, cash-poor firms—companies with a small amount of cash in the pre-Lehman period—decrease their output prices more than their counterparts, as shown later in Table IX. This result also holds at the industry level. As shown in Table VI. Panel A. column (3), when they face the negative credit supply shock, the firms in industries that lack internal liquidity decrease their prices more than their counterparts. Moreover, I find that the effect is stronger for the firms that had to pay out their loans in the post-Lehman period, as shown in Table VI, Panel A, column (4). Given that the Lehman failure was a surprise, firms that had to pay out their debts are likely to suffer more from financial problems and decrease their prices more to liquidate inventories and generate extra cash flow (Almeida et al. 2012). These analyses support the notion that firms sell off inventory to ensure their liquidity when they face an exogenous increase in the cost of external finance.

32. The credit supply shock interacted with firm-group-level initial sales is controlled for to compare the firms that generate similar sales. Similarly, for the industry-level regression, I controlled for the initial industry-level value of shipment. This specification is more general than the specification that uses the conventional measure of inventory-to-sales ratio, which restricts the elasticity of inventory on price to be the same as the negative elasticity of sales on price. As shown in Online Appendix S5, the importance of the raw material inventory suggests the need for such a generalization.

In addition, as shown in Table VI, Panel B, the effect of the credit supply shock is weaker for the firms that issued a bond or had multiple lead lenders in the pre-Lehman period. These results show that the bank shock is less damaging to the companies that can rely on an alternative source of financing or alternative banks. Moreover, consistent with the studies that find that the effect of a credit supply shock is larger for small companies, the effect is stronger for the firms that had smaller total assets and fewer employees (e.g., Gertler and Gilchrist 1994; Duygan-Bump, Levkov, and Montoriol-Garriga 2015). These results are consistent with previous studies and confirm the validity of the credit supply shock measure used in this article.

Finally, I explore the heterogeneous demand elasticity across firms and product groups and find that the decrease in output prices due to the negative credit supply shock is larger for the firms that face high demand elasticity, as shown in Table VI, Panel C. Allowing firm fixed effects does not alter this result, which suggests that among its various product categories, a firm chooses to decrease the price of products for which demand is more elastic. This result is intuitive, as firms would receive a larger cash flow from the product market by lowering their prices when they face more elastic demand.³³ The estimated elasticities in the regression analysis are based on the nested CES demand system in the Appendix, and the results are robust to different market structure assumptions.³⁴ Note that for Panel C, I used the utilitybased ideal price index along with the demand elasticities to coherently conduct the analyses within a nested CES demand framework. Other results are robust to using the utility-based price index.

33. Of course, such firms might have less incentive to lower their prices. If firms that face a negative bank shock aim to generate a particular amount of sales, the effect might be stronger for the firms that face an inelastic demand, as such firms have to decrease prices more to generate the targeted amount of sales. However, the reduced-form empirical results in this article do not support this prediction.

34. The derivation of demand elasticity under different market structure assumptions is in Appendix A of Hottman, Redding, and Weinstein (2016). I also tried the same regression with the HHI index and concentration ratio to understand how the effect differs across the degree of competition, but the estimated coefficients are not statistically significant enough to infer anything conclusive. I also measured and used the durability index but did not find statistically significant results.

V. RECONCILIATION WITH GILCHRIST ET AL. (2017)

The results in this article appear to be inconsistent with the results in Gilchrist et al. (2017), who conclude based on their empirical findings that financially constrained firms raise their output prices relative to the prices of their counterparts. I argue that my empirical analyses and the inventory mechanism in fact are fully consistent with their results that firms with a small amount of liquidity (or in a weak liquidity position) raise their output prices. The difference between these studies lies in the interpretation of the empirical results.

I first replicate and confirm their results in my sample by using their measure of financial constraint, liquidity:

(10)
$$\Delta \ln P_{fg} = \beta_0 + \eta_1 \text{LIQ}_f + \eta_2 X_f + \varepsilon_{fg},$$

where LIQ_f stands for liquidity, which is either contemporaneous (2008) cash to total assets or lagged (2006) cash to total assets. A fundamental difference relative to equation (6) is the measure of the credit supply shock. My baseline specification follows Gilchrist et al. (2017) closely and includes a lagged log change in sales, a lagged log change in the cost of goods sold, and the initial inventory-to-sales ratio as control variables (X_f). These controls aim to address the concerns related to the change in output prices, such as a change in demand, cost, and the inventory of firms. I additionally control for the lagged dependent variable, similar to their measure of the lagged industry-level inflation. Given the limited availability of some control variables, following the previous study, I restrict my sample to listed firms.

Even in my sample, the firms with a large amount of cash before or during the banking crisis lowered their output prices relative to the output prices of their counterparts. Table VII reports the results by using the lagged liquidity, which is perceived to be more exogenous compared to the contemporaneous liquidity position. Based on column (1), the coefficient of lagged liquidity position is negative and statistically significant at the conventional level, which confirms the results in Gilchrist et al. (2017). The same replication results hold with the contemporaneous liquidity position, as shown in Online Appendix S7. Column (2) additionally allows the measure of the bank shock, and in this specification, both the bank shock and liquidity independently explain the output price dynamics. These results confirm that the key difference

	Δln	$P_{\rm fg}: 2006 {\rm q4-2}$	2007q2 to 2008q4–	2009q2
		Includi	ng X_f related to	
	Gilchrist e	t al. (2017) (2)	Bates, Kahle, a (3)	and Stulz (2009) (4)
2006 LIQ _f	-2.84**	-2.17^{*}	0.43	0.04
$(-\Delta L_{\mathrm{f}})$	(1.40)	(1.21) -1.99^{**} (0.94)	(2.14)	(2.10) -3.37^{**} (1.40)
2006 CF volatility		(0.01)	-2.20^{**} (0.93)	-2.15^{***} (0.79)
Observations	947	947	947	947

TABLE VII

THE EFFECT OF CORPORATE LIQUIDITY ON OUTPUT PRICES

Notes. *p < .10, **p < .05, **p < .01; the standard errors are clustered by firm and product group, and the regression is weighted by initial sales. 2006 LIQ_f is the cash to assets in 2006, and 2006 CF volatility is defined as the standard deviation of cash flow to assets for the past 10 years. The set of firm-level controls related to Gilchrist et al. (2017) are the firm-level 2006 inventory to sales, the 2004–2006 change in market share at the firm-group-level, and the 2004–2006 change in the number of employees. The set of firm-level controls related to Bates, Kahle, and Stulz (2009) are the 2006 capital expenditure to assets, 2006 acquisitions to assets, and 2006 debt to assets. Across all specifications, the quality-adjusted utility-based price index is used, and the lagged dependent variable is included, similar to what had been done in Gilchrist et al. (2017), who use the quality-adjusted price index and control for the lagged industry-level inflation. All reported variables are normalized to have a unit variance to facilitate the comparison of coefficients.

in this article relative to the previous study is the measure of financial constraint, not the sample or regression specifications.³⁵

However, the effect of corporate liquidity is sensitive to the inclusion of other proxies of financial constraint, especially cash flow volatility. Since Bates, Kahle, and Stulz (2009) document the correlation between liquidity and other variables, which might confound the effect of liquidity on output price, column (3) considers an alternative set of control variables: initial cash flow volatility, capital expenditure to assets, acquisitions to assets, and debt to assets. In this specification, the coefficient of liquidity changes sign and loses the conventional level of statistical significance. In particular, the results suggest that the firms faced with larger cash flow volatility happened to hold more liquidity initially, and during the financial panic, they lowered their output prices; the cash flow volatility negatively affects price and takes over the

35. In addition, as shown in Online Appendix S6.H, the main result in this article is robust to using only listed firms under my preferred specification. Note that the timing of the event cannot explain the difference between the two studies. Consistent with this article, Gilchrist et al. (2017) focus on the period of Lehman Brothers' failure as shown in their figures and tables.

effect of the liquidity in columns (1) and (2). Column (4) allows the bank shock in addition to cash flow volatility and other controls. Although the bank shock and the cash flow volatility separately explain output price growth, the liquidity still has no meaningful effect on output prices. Online Appendix S7 considers a similar exercise with both listed and unlisted firms along with controls used in Table IV. The bank shock and liquidity affect output price dynamics without control variables, but after adding the controls, the coefficient of the bank shock becomes larger and remains statistically significant, as in Table IV, whereas the coefficient of initial liquidity changes sign and becomes statistically insignificant, as in Table VII.

The unstable estimated coefficient of corporate liquidity supports the view in the corporate finance literature that corporate liquidity is endogenously allocated across firms and cannot be used as a measure of financial constraint. In fact, by adopting corporate liquidity as a dependent variable in their regression analyses, a vast body of literature asks why companies hold liquidity. Almeida et al. (2014) survey this literature and conclude that firms hold more liquidity because they are more likely to be financially constrained. This argument dates back to Keynes (1936), who discusses that there is a fundamental relationship between corporate liquidity management and financial friction and emphasizes the precautionary saving motive to explain the variation in corporate liquidity position.

More specifically, consider the contemporaneous (2008) liquidity position. Kahle and Stulz (2013) find that the bankdependent firms that were likely to be more affected by a credit shortage raised—not lowered—their liquidity in 2008. I confirm this finding by showing that the companies that face a negative credit supply shock raise their liquidity, as shown in Table V. This result is consistent with the fire sale of inventory hypothesis and reflects that such firms want to sell their inventory to hold more liquidity due to the precautionary motive. However, such companies that face a negative bank shock would then be classified as firms in a strong liquidity position, not in a weak liquidity position.

Regarding the lagged (2006) liquidity position, although it cannot be an outcome of the 2007–2009 financial crisis, there is still the question of why some firms held more liquidity than their counterparts in the precrisis period. By using the 2006 liquidity, I replicate a seminal work by Bates, Kahle, and Stulz (2009) and

		Cas	h to assets	ratio	
	(1)	(2)	(3)	(4)	(5)
Cash flow volatility	0.47***				0.14***
·	(0.10)				(0.04)
Capital expenditure to assets		-1.33^{***}			-0.89^{***}
* *		(0.24)			(0.25)
Acquisition to assets			-0.71^{***}		-0.53^{***}
*			(0.13)		(0.06)
Debt to assets				-0.45^{***}	-0.40^{***}
				(0.05)	(0.04)
Firm-level controls	No	No	No	No	Yes
Observations	1,701	1,701	1,701	1,701	1,701

TABLE VIII

CORPORATE LIQUIDITY AND FIRM-LEVEL CHARACTERISTICS IN 2006

Notes. *p < .10, **p < .05, **p < .01; the standard errors are clustered by the two-digit SIC industry code. The firm-level controls are the two-digit SIC, firm size, market to book ratio, networking capital to assets, dividend dummy, and R&D to sales. The construction of the variables and the choice of control variables follow Bates, Kahle, and Stulz (2009) closely, as reported in Online Appendix S4.

find that the firms that have more liquidity are likely to be more financially constrained relative to their counterparts before the financial panic. Bates, Kahle, and Stulz (2009) identify more than 10 factors that lead firms to hold more liquid assets. In particular, they find that more cash holding (or a strong liquidity position) is associated with less investment, borrowing, acquisitions, and unstable cash flow, which are characteristics that likely reflect constrained companies rather than unconstrained companies. Table VIII confirms that their results hold in 2006, consistent with the results in Table VII. In Online Appendix S7, by reporting the same relationships between 2006 corporate liquidity and 2008 firm-level characteristics. I provide suggestive evidence that such firms remained constrained in the middle of the financial crisis. In this replication analysis, I use the Compustat database, which is used in Bates, Kahle, and Stulz (2009) and Gilchrist et al. (2017).³⁶ This result, along with previous studies, emphasizes that the concern about using liquidity as a measure of financial constraint is not specific to the particular sample that I use but generally applies to different data and periods. More generally, this concern

^{36.} I closely follow Bates, Kahle, and Stulz (2009) for cleaning the Compustat database. See Online Appendix S4 for a more detailed discussion of the Compustat data used in this analysis.

is consistent with the findings in Farre-Mensa and Ljungqvist (2016), who show that the measures of financial constraint made with firm-level balance sheet variables do not correctly identify the true level of financial constraint.

Instead of using liquidity as an exogenous explanatory variable, to improve the identification strategy used in equation (10) I include the measure of the bank shock and interact it with the 2006 liquidity, as in equation (9). Although the discussion of the previous literature and the regression analyses in this section show the drawbacks of using the cash to assets ratio as a financial shock, simultaneously, it could be a good measure of responsiveness to the bank shock. Based on previous studies, Tables VII and VIII, and the results in Online Appendix S7, it is likely that financially constrained firms choose to hold a larger amount of liquidity before the financial crisis and were kept more constrained in the Great Recession than their counterparts. However, as a result of holding more internal liquidity, all else being equal, such firms would be less sensitive to the drying up of external funds.³⁷ By adding and interacting the bank shock with the initial liquidity, I seek to stress the benefit of liquidity, which is the mitigation of the effect of the external credit crunch, rather than other characteristics of liquidity. This identification strategy is similar to the strategy in Jeenas (2019), who uses the interaction of the firmlevel initial cash holding and the exogenous aggregate monetary policy shock.³⁸

As reported in Table IX, column (1), although firms decrease their output prices when they face a negative bank shock, the

37. Note that such a hedging role of cash holding is fully consistent with the interpretation of cash holdings in Gilchrist et al. (2017) and in other papers that use the initial cash holding as an independent variable, such as Bacchetta, Benhima, and Poilly (2019) and Melcangi (2019).

38. Another way to think about equation (10) is that initial liquidity is used as a measure of the responsiveness to the 2008–2009 financial crisis. However, to use the 2008–2009 financial crisis as a shock, one needs to make a strong assumption that other events in the same period did not differentially affect the firms that have more initial liquidity. For example, if a fall in the aggregate demand or a rise in the aggregate uncertainty in the same period differentially affects the risky firms that had high cash flow volatility, the coefficient of initial liquidity would be biased, as shown in Table VII, because liquidity and cash flow volatility are positively correlated, as shown in Table VIII. Such narratives potentially explain why, as shown in Online Appendix S7, the firms that have more liquidity in 2006 were likely to remain constrained during the financial crisis. Instead of using 2008 as a financial shock, equation (9) adapts the bank shock to separate the external financial shock from other events that happened in 2008.

	$\Delta { m ln} ilde{P}_{ m fg}$: 2006q4–200	07q2 to 2008	3q4-2009q2
	Neg	gative bank s	hock measu	red with
	$(-\Delta L_f)$ (1)	Lehman (2)	ABX (3)	(-BankItem) (4)
Negative bank shock \times LIQ _f	1.14***	5.10***	2.28^{**}	0.63***
	(0.31)	(0.65)	(1.14)	(0.10)
Negative bank shock	-12.85^{**}	-18.69^{***}	-19.27^{**}	-9.73^{***}
	(5.99)	(2.42)	(9.20)	(1.40)
Firm-level controls	Yes	Yes	Yes	Yes
Product group FE	Yes	Yes	Yes	Yes
Observations	832	832	832	832

TABLE IX

TREATMENT INTERACTION: BANK SHOCK AND CORPORATE LIQUIDITY

Notes. *p < .10, **p < .05, **p < .01; the standard errors are clustered by firm and product group; and the regression is weighted by initial sales. The firm-level controls are the firm's listed status, four-digit NAICS fixed effects, age, size, bond rating, number of loans, amount of loans, loan type, loan-year fixed effects, multi-lead fixed effects, number of loans due in the post-Lehman period fixed effects, loan spread, and loan maturity. The Lehman exposure, ABX securities exposure, and bank items are used as direct measures of the bank shock. Similar to my treatment of the main leave-one-out credit supply shock measure, for consistency, I change the sign of the bank items. All four bank shock measures are standardized to have a unit variance. The 2006 cash to assets variable in percentage points is used to ease the interpretation.

effect is smaller when firms have more initial liquidity. An increase by 1 percentage point of the 2006 cash to assets ratio mitigates the effect of the bank shock by approximately 1.14%. The qualitative result is robust to using three alternative measures of bank shocks, as shown in columns (2)-(4) of the same table. This result also clarifies how the main results in this article are related to the results in Gilchrist et al. (2017). The unconditional effect of initial corporate liquidity on output prices might be negative, as in Table VII, because financially constrained firms prefer to hold more initial liquidity and lower their output prices and inventory stock relative to their counterparts. However, conditioning on the negative bank shock, as shown in Table IX, the effect of liquidity on the output price is positive, because more initial liquidity can ironically benefit firms in alleviating the external financial stress in the middle of a financial crisis. These results are consistent with the interpretation that financially constrained firms decrease their output prices.³⁹

39. Another way to modify equation (10) to address the endogeneity concern is to include more control variables, which are likely to correlate with the 2006 liquidity and simultaneously affect the output price during the Great Recession. Moreover, Table IX provides support to the inventory adjustment hypothesis. Before the financial panic, companies originally had two sources of liquidity to manage their operations: internal liquidity (cash) and external liquidity (banks). When there is a surprising increase in the external cost of funding, the firms that had a lot of cash at the beginning of the period would not need to sell off their inventory and decrease their price to generate extra cash. However, the firms that lack internal liquidity are more likely to sell their inventories and decrease their output prices to ensure extra funds. The results emphasize the rich interaction of output price, corporate inventory, and corporate liquidity in the middle of the financial crisis.

VI. Aggregate Implications

To provide a back-of-the-envelope calculation for the effect of the credit supply shock on aggregate inflation dynamics through the inventory adjustment mechanism, this section presents a streamlined discussion of a simple business cycle model that relates the micro-level estimates to aggregate inflation dynamics. Although the reduced-form micro-level regression framework with bank shocks is useful for identifying the credit supply shock with a minimal number of assumptions, these results can speak only to a relative change in the variable dynamics of interest because of the framework's reliance on the cross-sectional variation in the data. To analyze the aggregate dynamics, I include in the model two identical groups of producers that face different degrees of a credit supply shock. This formulation allows me to take advantage of micro-level empirical evidence to calibrate the parameters in the model, and through the lens of the model, I

However, given that there are many variables that are correlated with the liquidity position as shown in Bates, Kahle, and Stulz (2009), it might be difficult to generate enough variation in this variable to consistently estimate the effect of liquidity when all of these variables are controlled. Indeed, in my sample, when I include 2006 liquidity and other variables in the regression, as shown in Table VII and Online Appendix S7, the estimated coefficient of liquidity loses statistical significance. Moreover, a bigger concern is the unobserved characteristics that cannot be easily measured in the data but potentially bias the effect of the 2006 liquidity on output prices. Note that I cannot use the credit supply shock as an instrumental variable for the liquidity measure because of the violation of an exogeneity condition. Under the inventory adjustment hypothesis, the credit supply shock does not only affect the output price through corporate liquidity.

address the aggregate variable dynamics.⁴⁰ Online Appendix S1 illustrates the model in detail along with other simpler models that formalize the relationships among output prices, inventory, and the credit supply shock.⁴¹

There are five types of agents: households, retailers, a central bank, and two otherwise identical representative entrepreneurs that face different degrees of credit supply shocks. With the exception of the entrepreneurs, who are counterparts to the producers studied extensively in the micro-level data, the modeling of agents follows the canonical monetary business cycle model presented in Iacoviello (2005). Both the households and the entrepreneurs maximize their lifetime utility subject to their budget constraint. The households consume the final goods purchased from the retailers, lend to the entrepreneurs, work for the entrepreneurs, and earn wages in compensation. The entrepreneurs consume the final goods purchased from the retailers, borrow from the households, and produce the intermediate goods by hiring the households and using their capital stocks. The retailers use what they purchase from the entrepreneurs, differentiate the products, and sell to the consumers. In this process, the retailers face Calvo-Yun price rigidity in changing their output prices. The central bank is assumed to follow the standard Taylor rule in the main body of the article, but Online Appendix S1.B discusses the amplifying role of the zero-lower bound observed in the Great Recession.42

There are two fundamental features of the entrepreneurs that aim to match the micro-level empirical evidence. First, to integrate the fire sale of inventory hypothesis, I adapt the product stock-out motive of inventory holding, as described in

40. This framework is similar to the framework of Nakamura and Steinsson (2014), who also exploit cross-sectional variation to estimate the key parameter and relate it to the aggregate variable (multiplier) by using the business cycle model.

41. Online Appendix S1.A shows a simpler version of the model that captures the dynamics of the relative variables without price rigidity and a central bank, and Online Appendix S1.C shows an analytical example of the partial equilibrium model of (S,s) inventory holding presented in Alessandria, Kaboski, and Midrigan (2010b).

42. To motivate a zero lower bound, I fix the interest rate for four quarters and then allow it to follow the Taylor rule. This analysis makes the effect of the decrease in the borrowing capability of the entrepreneur on aggregate inflation even stronger since the central bank cannot use monetary policy to counteract the deflationary force generated by the inventory adjustment mechanism. Wen (2011).⁴³ I assume that entrepreneurs produce a continuum of products and that each product faces an idiosyncratic shock. The shock is realized after entrepreneurs produce their products, and this timing lag gives them an incentive to store products in inventory to avoid product stock-outs.

Second, to mimic the negative credit supply shock in the empirical analysis, entrepreneurs face a borrowing capability that is exogenously given to them. The thought experiment is the following: given a sudden decrease in one representative entrepreneur's borrowing capacity, how do the output price and inventory dynamics of one entrepreneur evolve compared with that of the other entrepreneur? Once I calibrate the magnitude of the change in the entrepreneur's financial capability based on the output price's relative changes, which is 15%, I use the same shock to understand the behavior of aggregate inflation and inventory dynamics.⁴⁴ All other parameters in the model are calibrated to match the aggregate moments following the literature and are documented in Online Appendix S1.A.⁴⁵ Note that consistent with the empirical results in Table II, entrepreneurs do not hold inventory in preparation for the borrowing constraint in the model.⁴⁶

The key intuition of the model is the consumption-smoothing motive of a representative entrepreneur that faces a decrease in

43. One way to think about the model is the integration of the stock-out avoidance motive of inventory holding developed by Wen (2011) into a simple version of the business cycle model presented in Iacoviello (2005).

44. To match the model with the data, I define an indicator variable that equals 1 if the measure of the credit supply shock is larger than the 50th percentile of the measure and equals 0 otherwise. I regress the change in log price on this measure and find that the effect of this indicator variable on output price is approximately 15%, as documented in Online Appendix S8.

45. Note that I do not match the parameters based on the consumer packaged goods because the objective of the model is to look at the aggregate dynamics. The external validity exercises reported in Online Appendix S6.K justify looking at aggregate dynamics with the model. In addition, the movements in the aggregate series in this period, such as changes in prices, inventory, corporate cash holdings, and the credit condition, are consistent with the inventory mechanism and additionally support the generalization of the inventory mechanism.

46. Requiring entrepreneurs to hold inventory to avoid a credit crunch would likely amplify the inventory adjustment mechanism because there is more incentive for entrepreneurs to liquidate inventory when they face a negative credit supply shock.

her borrowing capability.⁴⁷ Such constrained entrepreneurs face a decrease in consumption in this period due to the tightening of the borrowing constraint. To smooth their consumption, they sell off their inventory at a low price and generate extra sales from the product market. However, subsequently, they run out of inventory and increase their output prices. This behavior generates the temporary decrease in relative output prices and inventory observed in Figure III and Table V, as shown in Online Appendix S1.B.

Once I calibrate the magnitude of the relative change in borrowing capability based on the micro-level data. I look at how much the effect of such relative changes can explain the aggregate inflation dynamics in this canonical monetary business cycle model. To understand the counterfactual scenario where entrepreneurs do not decrease their inventory. I change the capital adjustment cost to evaluate the prediction without changing the steady state. If the capital adjustment cost is high, entrepreneurs sell off their inventory when they face a decrease in borrowing capability. However, when there is no capital adjustment cost, entrepreneurs instead decrease their capital investment and use these resources to increase their present consumption. Without a capital adjustment cost, it is more efficient for entrepreneurs to decrease their capital investment instead of their inventory stock because the decrease in inventory raises the probability of a product stock-out.

Figure V presents the predicted aggregate inflation dynamics with respect to the exogenous decrease in the borrowing capability of one type of entrepreneur in the model along with the U.S. inflation dynamics observed in the data in the middle of the financial panic. In the benchmark case where entrepreneurs liquidate inventory to generate additional sales from the product market, when there is a decrease in the borrowing capacity of one of two representative entrepreneurs, there is an approximately 9% decrease in the aggregate inflation in the model followed by a similar increase in inflation. Considering the observed aggregate

^{47.} For the listed companies, one can think of the dividend-smoothing motive of producers, which is well documented in the studies of corporate financial policy (e.g., Brav et al. 2005). Bianchi and Bigio (2018) also use such interpretations in modeling the consumption of bankers. Instead of using the consumption-smoothing motive, it is likely that one can generate the inventory adjustment with respect to a financial shock by explicitly integrating the bankruptcy of producers.





The Aggregate Inflation Dynamics in the Model and Data

Panel A shows the predicted dynamics of aggregate inflation due to the negative credit supply shock to one of two representative entrepreneurs in the model. The solid line reflects the aggregate inflation dynamics when producers sell their inventory to alleviate the instead of inventory to alleviate the credit supply shock. Panel B shows the U.S. inflation dynamics observed in the data during the effect of the credit supply shock, and the dotted line is the aggregate inflation dynamics when producers decrease their capital investment financial panic, which is calculated from the producer price index (PPIACO_PCH) made available by the Bureau of Labor Statistics. inflation in the data, such a borrowing shock can capture approximately 80% of the decrease in the short-run aggregate inflation in this period. However, subsequently, there is a large increase in aggregate inflation in the model, as entrepreneurs run out of inventory. Such a rise in inflation resembles the prediction of influential macro models that explain the stable inflation in this period despite a large fall in aggregate demand (e.g., Del Negro, Giannoni, and Schorfheide 2015; Christiano, Eichenbaum, and Trabandt 2015; Gilchrist et al. 2017). To understand the stable inflation dynamics in this period, these models integrate the inflationary pressure of a financial shock through various mechanisms other than the inventory adjustment. In contrast, the model presented in this article suggests that the inventory adjustment due to a similar financial shock generates inflationary pressure that follows the deflationary force. When entrepreneurs decrease capital investment instead of adjusting inventory, the financial shock has a negligible effect on the inflation dynamics because entrepreneurs do not need to lower their output prices to decrease their capital investment.

Overall, the simple business cycle model in this article emphasizes the importance of inventory by providing a rough estimate for the effect of the credit supply shock on aggregate inflation dynamics with and without an inventory adjustment mechanism. The results should be interpreted with caution since the inflation in this period is affected by many other factors that are not explicitly modeled.⁴⁸ The impulse response function presented in Figure V should be interpreted as a back-of-the-envelope calculation for the effect of a financial shock on aggregate inflation dynamics with and without an inventory adjustment mechanism.

VII. CONCLUSION

In this article, by using novel micro-level data and a change in bank health at the time of the Lehman Brothers failure as an exogenous variation of companies' credit condition, I find that the firms facing a negative credit supply shock decrease their output prices. I posit a "fire sale" of inventory hypothesis to explain this

^{48.} For example, a fall in housing prices, a rise in uncertainty, a fall in international trade, and a fall in oil and commodity prices are not present in the model. Other features, such as the input-output production network and industry and firm heterogeneity in inventory holding, would also affect the quantification.

empirical finding: firms that face a negative credit supply shock decrease their prices because they need to quickly sell off their inventories and generate extra cash. I empirically support this hypothesis by using macro-level and micro-level data.

This study reveals that the corporate inventory used for liquidity management, which has been neglected in previous studies of banking crises, is a crucial determinant of output price dynamics. Models that feature inventories will better account for the fluctuation of output prices, inventory, and other economic and financial variables.

Appendix

This appendix presents the results based on the utility-based price index. I adapt the nested CES demand system in Hottman, Redding, and Weinstein (2016) to build the firm-group-specific price index from the ACNielsen Homescan Panel database.

Consider the following Cobb-Douglas utility function:

(11)
$$\ln \mathbb{U}_t = \int_{g \in \Omega} (\varphi_{gt} \ln C_{gt}) dg, \int_{g \in \Omega} \varphi_{gt} dg = 1,$$

where subscript g is the product group, and t is time. Ω is the set for a product group, and φ_{gt} is a consumer's perceived product group quality (or appeal/taste) at time t. C_{gt} is the group-time-specific consumption index that corresponds to the following CES nests:

(12)

$$C_{gt} = \left[\sum_{f \in \Omega_{gt}} \left(\varphi_{fgt} C_{fgt}\right)^{\frac{\sigma_g^F - 1}{\sigma_g^F}}\right]^{\frac{\sigma_g^F}{\sigma_g^F - 1}}, C_{fgt} = \left[\sum_{u \in \Omega_{fgt}} \left(\varphi_{ut} C_{ut}\right)^{\frac{\sigma_g^U - 1}{\sigma_g^U}}\right]^{\frac{\sigma_g^U}{\sigma_g^U - 1}},$$

where subscript f is the firm and u is the UPC or barcode-level product, Ω_{gt} is the set of the firms within product group g at time t, Ω_{fgt} is the set of the UPCs made by firm f in product group gat time t, φ_{fgt} captures perceived firm-group quality at time t, φ_{ut} captures the perceived UPC quality made by firm f in product group g at time t, σ_g^F governs the elasticity of substitution across firms for each product group, and σ_g^U governs the elasticity of

substitution across UPCs for each firm.⁴⁹ Under this structure, firm appeal (φ_{fgt}^F) affects the sales of all products supplied by firm f within product group g proportionately, while product appeal (ϕ_{ut}^U) determines the relative sales of individual products u within firm f and product group g.

It is useful to illustrate the underlying consumer behavior with the nested CES demand system used in this article. When consumers visit a store, the demand system assumes that they first decide which product group they will buy from, then decide which brand or firm's product to purchase, and then purchase a specific UPC. For example, a consumer decides to purchase jams, jellies, or spreads (product group), then decides to buy a Smucker's product (firm), and then chooses Smucker's sugar-free strawberryflavor fruit spread (UPC). The elasticities govern how sensitively consumers react to changes in the output price, and the perceived quality parameters govern how purchasing behavior is affected by factors other than output prices, such as product quality (e.g., organic versus nonorganic), brand quality, and product/brand advertisement.

The corresponding well-known exact CES price indices are

(13)

$$P_{gt} = \left[\sum_{f \in \Omega_{gt}} \left(\frac{P_{fgt}}{\varphi_{fgt}}\right)^{1 - \sigma_g^F}\right]^{\frac{1}{1 - \sigma_g^F}}, P_{fgt} = \left[\sum_{u \in \Omega_{fgt}} \left(\frac{P_{ut}}{\varphi_{ut}}\right)^{1 - \sigma_g^U}\right]^{\frac{1}{1 - \sigma_g^U}}$$

and the expenditure shares of products are⁵⁰

(14)
$$S_{fgt} = \frac{\left(\frac{P_{fgt}}{\varphi_{fgt}}\right)^{1-\sigma_g^F}}{\sum_{k\in\Omega_{gt}}\left(\frac{P_{kgt}}{\varphi_{kgt}}\right)^{1-\sigma_g^F}}, S_{ut} = \frac{\left(\frac{P_{ut}}{\varphi_{ut}}\right)^{1-\sigma_g^U}}{\sum_{k\in\Omega_{fgt}}\left(\frac{P_{kt}}{\varphi_{kt}}\right)^{1-\sigma_g^U}}.$$

49. As discussed in Hottman, Redding, and Weinstein (2016), φ_{fgt}^{F} cannot be defined independently of φ_{ut}^{U} because the utility is homogeneous of degree one in perceived firm quality. I normalize the quality parameter: $\tilde{\varphi}_{gt}^{F} = \left(\prod_{f \in \Omega_{gt}^{U}} \varphi_{fgt}^{T}\right)^{\frac{1}{N_{gt}^{F}}} = 1$, $\tilde{\varphi}_{fgt}^{U} = \left(\prod_{u \in \Omega_{fgt}^{U}} \varphi_{ut}^{U}\right)^{\frac{1}{N_{fgt}^{U}}} = 1$, where N_{gt}^{F} is the number of firms in product group g at time t and N_{fgt}^{U} is the number of UPCs made by firm f within group g at time t. 50. Equation (14) can be recovered by using Shephard's lemma.

	$\Delta \ln P_{\rm fg}$: 2006q4–2007q2 to 2008q4–2009q2						
	OLS		$(-\Delta L_f)$ instrumented using				
			Lehman	ABX	BankItem	All	
	(1)	(2)	(3)	(4)	(5)	(6)	
$(-\Delta L_f)$	-2.25^{***}	-8.57^{***}	-7.32^{***}	-7.00^{***}	-8.80***	-7.88^{***}	
,	(0.76)	(1.46)	(2.66)	(2.53)	(3.01)	(1.73)	
Firm-level controls	No	Yes	Yes	Yes	Yes	Yes	
Product group FE	No	Yes	Yes	Yes	Yes	Yes	
NAICS four-digit FE	No	Yes	Yes	Yes	Yes	Yes	
First-stage F statistic			16.80	7.80	14.90	11.90	
J-statistic p -value						.89	
$E[\Delta \ln P]$	11.379	11.4	11.4	11.4	11.4	11.4	
$E[\Delta \ln P: \Delta L_{p90} - \Delta L_{p10}]$	-4.916	-18.7	-16	-15.3	-19.2	-17.2	
Observations	1,658	1,658	1,658	1,658	1,658	1,658	

TABLE A.1								
MAIN RESULTS	WITH THE	UTILITY-BASED	PRICE	INDEX				

Notes. *p < .10, **p < .05, ***p < .01; the standard errors are clustered by firm and product group, the regression is weighted by initial sales, and the firm-level controls are the firm's listed status, age, bond rating, number of loans, amount of loans, loan type, loan-year fixed effects, multi-lead fixed effects, number of loans due in the post-Lehman period fixed effects, loan spread, loan mutrity, and lagged $AlnP_{fir}$.

Equation (14) clarifies how this framework perceives UPCspecific and firm-specific qualities, φ_{ut} and φ_{fgt} . These qualities change the market share holding the output price constant. If two products have the same price but one has a larger market share, then this product has a higher perceived quality.

The relative market share can be derived from equation (14):

(15)
$$\frac{S_{ut}}{\tilde{S}_{fgt}} = \frac{\left(\frac{P_{ut}}{\varphi_{ut}}\right)^{1-\sigma^U}}{\left(\frac{\tilde{P}_{ut}}{\tilde{\varphi}_{ut}^U}\right)^{1-\sigma^U}},$$

where $\tilde{S}_{fgt} = \left[\prod_{u \in \Omega_{fgt}} S_{ut}\right]^{\frac{1}{N_{fgt}}}$, which is the geometric average of the market share of UPCs for firm f within group g at time t. By plugging equation (15) into equation (13), one can derive the following firm-group-time price index

(16)
$$\ln P_{fgt} = \underbrace{\ln \tilde{P}_{fgt}}_{\text{Standard Index}} - \underbrace{\frac{1}{\sigma_g^U - 1} \ln \left[\sum_{u \in \Omega_{fgt}} \frac{S_{ut}}{\tilde{S}_{fgt}}\right]}_{u \in \Omega_{fgt}},$$

where the first term is the geometric average of the UPC-level price within the firm and the group, which is used in the main body of the article. This term is analogous to the standard price index, such as the Törnqvist or Laspeyres index. The second term is a variant of the Theil index, which measures quality and variety correction in the price index. Note that $\sum_{u} \frac{S_{ut}}{S_{fat}}$ in the second term increases if (i) the number of UPCs by firm *f* within group *g* (N_{fgt}) increases (variety effect), or (ii) the UPC share dispersion within the firm increases (quality effect).⁵¹

To measure the price index in equation (16), I use the estimated demand elasticities (σ_g^U) from Hottman, Redding, and Weinstein (2016).⁵² Identical to the main analysis, I take a geometric average across quarters within 2006:Q4–2007:Q2 (the last three quarters in the pre-Lehman period) and 2008:Q4–2009:Q2 (the post-Lehman period) to make the price index comparable to the credit supply shock. I then take the difference of the logged price index across pre- and post-Lehman periods.

Table A.1 presents the results that replicate Table IV with the utility-based firm-group-specific price index in equation (16). All the coefficients are negative and statistically significant, and the magnitudes of the estimated coefficients are similar to the magnitudes of the estimated coefficients in Table IV. The results are robust to other price indices, as reported in Online Appendix S6.G.

JOHNS HOPKINS UNIVERSITY

SUPPLEMENTARY MATERIAL

An Online Appendix for this article can be found at *The Quar*terly Journal of Economics online.

51. The UPC share dispersion reflects the perceived product quality. For example, suppose that consumers see two products offered by the same firm at the same price. It is better for consumers to see one high-quality product and one low-quality product rather than two mediocre products because one can always choose a high-quality product in the former scenario, whereas they must always choose a mediocre product in the other scenario. This intuition is reflected in the share dispersion term, which measures heterogeneity in product quality.

52. Hottman, Redding, and Weinstein (2016) apply a modification of the "identification through heterogeneity" method originally developed by Feenstra (1994) to the same data set that I use in this article. I am grateful to the authors for providing these estimates.

DATA AVAILABILITY

Code replicating the tables and figures in this article can be found in Kim (2020) in the Harvard Dataverse, doi: 10.7910/DVN/67Y8VE.

References

- Alessandria, George, Joseph P. Kaboski, and Virgiliu Midrigan, "The Great Trade Collapse of 2008–09: An Inventory Adjustment?," *IMF Economic Review*, 58 (2010a), 254–294.
 - ——, "Inventories, Lumpy Trade, and Large Devaluations," *American Economic Review*, 100 (2010b), 2304–2339.
- ——, "U.S. Trade and Lumpy Inventory Dynamics," *American Economic Review: Papers and Proceedings*, 101 (2011), 303–307.
- ——, "Trade Wedges, Inventories, and International Business Cycles," Journal of Monetary Economics, 60 (2013), 1–20.
- Almeida, Heitor, Murillo Campello, Igor Cunha, and Michael Weisbach, "Corporate Liquidity Management: A Conceptual Framework and Survey," Annual Review of Financial Economics, 6 (2014), 135–162.
- Almeida, Heitor, Murillo Campello, Bruno Laranjeira, and Scott Weisbenner, "Corporate Debt Maturity and the Real Effects of the 2007 Credit Crisis," *Critical Finance Review*, 1 (2012), 3–58.
- Amiti, Mary, and David E. Weinstein, "Exports and Financial Shocks," Quarterly Journal of Economics, 126 (2011), 1841–1877.
 - ——, "How Much Do Idiosyncratic Bank Shocks Affect Investment? Evidence from Matched Bank-Firm Loan Data," *Journal of Political Economy*, 126 (2018), 525–587.
- Anderson, Simon, Andre de Palma, and Jacques-Francois Thisse, *Discrete Choice Theory of Product Differentiation* (Cambridge, MA: MIT Press, 1992).
- Aragon, George O., and Philip E. Strahan, "Hedge Funds as Liquidity Providers: Evidence from the Lehman Bankruptcy," *Journal of Financial Economics*, 103 (2012), 570–587.
- Atkeson, Andrew, and Ariel Burstein, "Pricing to Market, Trade Costs and International Relative Prices," American Economic Review, 98 (2008), 1998-2031.
- Bacchetta, Philippe, Kenza Benhima, and Celine Poilly, "Corporate Cash and Employment," American Economic Journal: Macroeconomics, 11 (2019), 30–66.
- Bartelsman, Éric J., Randy A. Becker, and Gray Wayne, "The NBER-CES Manufacturing Industry Database," NBER, 2000.
 Barth, Marvin J. III, and Valerie A. Ramey, "The Cost Channel of Monetary Trans-
- Barth, Marvin J. III, and Valerie A. Ramey, "The Cost Channel of Monetary Transmission," NBER Macroeconomics Annual 2001, vol. 16, Ben S. Bernanke and Kenneth Rogoff, eds. (Cambridge, MA: MIT Press, 2002), 199–240.
 Bates, Thomas, Kathleen Kahle, and Stulz Rene, "Why Do U.S. Firms Hold So
- Bates, Thomas, Kathleen Kahle, and Stulz Rene, "Why Do U.S. Firms Hold So Much More Cash than They Used To?," *Journal of Finance*, 64 (2009), 1985– 2021.
- Becker, Bo, and Victoria Ivashina, "Cyclicality of Credit Supply: Firm Level Evidence," Journal of Monetary Economics, 62 (2014), 76–93.
- Bianchi, Javier, and Saki Bigio, "Banks, Liquidity Management and Monetary Policy," Working Paper, 2018.
- Bils, Mark, and James A. Kahn, "What Inventory Behavior Tells Us about Business Cycles," American Economic Review, 90 (2000), 458–481.
- Borenstein, Severin, and Nancy L. Rose, "Bankruptcy and Pricing Behavior in U.S. Airline Markets," American Economic Review, 85 (1995), 397–402.
- Brav, Alon, John R. Graham, Campbell R. Harvey, and Roni Michaely, "Payout Policy in the 21st Century," *Journal of Financial Economics*, 77 (2005), 483– 527.

- Burstein, Ariel, and Gita Gopinath, "International Prices and Exchange Rates," Handbook of International Economics, 4th ed. Gita Gopinath, Elhanan Helpman and Kenneth Rogoff, eds. (Amsterdam: Elsevier, 2014), 391–451.
- Campello, Murillo, Erasmo Giambona, John R. Graham, and Campbell R. Harvey, "Liquidity Management and Corporate Investment During a Financial Crisis," Review of Financial Studies, 24 (2011), 1944–1979.
- Carey, Mark, and Mark Hrycay, "Credit Flow, Risk, and the Role of Private Debt in Capital Structure," Federal Reserve Board Working Paper, 1999.
- Carpenter, Robert E., Steven M. Fazzari, and Bruce C. Petersen, "Inventory Investment, Internal-Finance Fluctuations, and the Business Cycle," Brookings Papers on Economic Activity, 25 (1994), 75-138.
 - , "Financing Constraints and Inventory Investment: A Comparative Study with High-Frequency Panel Data," Review of Economics and Statistics, 80 (1998), 513-519.
- Chava, Sudheer, and Michael R. Roberts, "How Does Financing Impact Investment? The Role of Debt Covenants," Journal of Finance, 63 (2008), 2085-2121.
- Chevalier, Judith A., "Capital Structure and Product-Market Competition: Empirical Evidence from the Supermarket Industry," American Economic Review, 85 (1995a), 415-435.

"Do LBO Supermarkets Charge More? An Empirical Analysis of the Effects

- of LBOs on Supermarket Pricing," Journal of Finance, 50 (1995b), 1095–1112. Chevalier, Judith A., and David S. Scharfstein, "Liquidity Constraints and the Cyclical Behavior of Markups," American Economic Review, 85 (1995), 390– 396.
 - -, "Capital-Market Imperfections and Countercyclical Markups: Theory and Evidence," American Economic Review, 86 (1996), 703–725.
- Chodorow-Reich, Gabriel, "The Employment Effects of Credit Market Disruptions: Firm Level Evidence from the 2008-09 Financial Crisis," Quarterly Journal of Economics, 129 (2014), 1-59.
- Christiano, Lawrence J., Martin S. Eichenbaum, and Mathias Trabandt, "Understanding the Great Recession," American Economic Journal: Macroeconomics, 7 (2015), 110-167.
- Crouzet, Nicolas, and Hyunseung Oh, "What Do Inventories Tell Us about News Driven Business Cycles?," Journal of Monetary Economics, 79 (2016), 49-66.
- Darmouni, Olivier, "Estimating Informational Frictions in Sticky Relationships," Journal of Finance, (forthcoming).
- Del Negro, Marco, Marco P. Giannoni, and Frank Schorfheide, "Inflation in the Great Recession and New Keynesian Models," American Economic Journal: Macroeconomics, 7 (2015), 168–196.
- Duygan-Bump, Burcu, Alexey Levkov, and Judit Montoriol-Garriga, "Financing Constraints and Unemployment: Evidence from the Great Recession," Journal of Monetary Economics, 75 (2015), 89-105.
- Eaton, Jonathan, Samuel Kortum, Brent Neiman, and John Romalis, "Trade and the Global Recession," American Economic Review, 106 (2016), 3401-3438.
- Edmond, Chris, Virgiliu Midrigan, and Daniel Yi Xu, "Competition, Markups, and the Gains from International Trade," American Economic Review, 105 (2015), 3183-3221.
- Fahlenbrach, Rudiger, Robert Prilmeier, and Rene Stulz, "This Time Is the Same: Using Bank Performance in 1998 to Explain Bank Performance during the Recent Financial Crisis," Journal of Finance, 67 (2012), 2139-2185.
- Farre-Mensa, Joan, and Alexander Ljungqvist, "Do Measures of Financial Con-straints Measure Financial Constraints?," Review of Financial Studies, 29 (2016), 271-308.
- Feenstra, Robert C., "New Product Varieties and the Measurement of International Prices," American Economic Review, 84 (1994), 157-177.
- Fisher, Jonas D. M., and Andreas Hornstein, "(S, s) Inventory Policies in General Equilibrium," Review of Economic Studies, 67 (2000), 117–145.
- Garcia-Macia, Daniel, and Alonso Villacorta, "Macroprudential Policy with Liguidity Panics," Working Paper, 2019.

- Gertler, Mark, and Simon Gilchrist, "Monetary Policy, Business Cycles, and the Behavior of Small Manufacturing Firms," *Quarterly Journal of Economics*, 109 (1994), 309–340.
- Gilchrist, Simon, Raphael Schoenle, Jae W. Sim, and Egon Zakrajsek, "Inflation Dynamics during the Financial Crisis," *American Economic Review*, 107 (2017), 785–823.
- Greenstone, Michael, Alexandre Mas, and Hoai-Luu Nguyen, "Do Credit Market Shocks affect the Real Economy? Quasi-Experimental Evidence from the Great Recession and 'Normal' Economic Times," *American Economic Journal: Economic Policy*, 12 (2020), 200–225.
- Hamilton, James, "Causes and Consequences of the Oil Shock of 2007–08," Brookings Papers on Economic Activity, 40 (2009), 215–259.
- Hottman, Colin, Steve Redding, and David Weinstein, "Quantifying the Sources of Firm Heterogeneity," *Quarterly Journal of Economics*, 131 (2016), 1291–1364.
- Iacoviello, Matteo, "House Prices, Borrowing Constraints, and Monetary Policy in the Business Cycle," *American Economic Review*, 95 (2005), 739–764.
- Ivashina, Victoria, and David Scharfstein, "Bank Lending during the Financial Crisis of 2008," Journal of Financial Economics, 97 (2010), 319–338.
- Jeenas, Priit, "Firm Balance Sheet Liquidity, Monetary Policy Shocks, and Investment Dynamics," Working Paper, 2019.
- Kahle, Kathleen M., and Rene M. Stulz, "Access to Capital, Investment, and the Financial Crisis," *Journal of Financial Economics*, 110 (2013), 280–299.
 Kalemli-Ozcan, Sebnem, Bent Sorensen, Carolina Villegas-Sanchez, Vadym
- Kalemli-Ozcan, Sebnem, Bent Sorensen, Carolina Villegas-Sanchez, Vadym Volosovych, and Sevcan Yesiltas, "How to Construct Nationally Representative Firm Level Data from the ORBIS Global Database," NBER Working Paper No. 21558, 2015.
- Kaplan, Steven N., and Luigi Zingales, "Do Investment-Cash Flow Sensitivities Provide Useful Measures of Financing Constraints?," *Quarterly Journal of Economics*, 115 (1997), 707–712.
- Kashyap, Anil K., Owen A. Lamont, and Jeremy C. Stein, "Credit Conditions and the Cyclical Behavior of Inventories," *Quarterly Journal of Economics*, 109 (1994), 565–592.
- Keynes, John M., The General Theory of Employment, Interest and Money (London: Macmillan, 1936).
- Khan, Aubhik, and Julia K. Thomas, "Inventories and the Business Cycle: An Equilibrium Analysis of (S, s) Policies," *American Economic Review*, 97 (2007), 1165–1188.
- Khwaja, Asim Ijaz, and Atif Mian, "Tracing the Impact of Bank Liquidity Shocks: Evidence from an Emerging Market," *American Economic Review*, 98 (2008), 1413–1442.
- Kim, Ryan, "Replication Data for: The Effect of the Credit Crunch on Output Price Dynamics: The Corporate Inventory and Liquidity Management Channel," (2020), Harvard Dataverse, doi: 10.7910/DVN/67Y8VE.
- Kryvtsov, Oleksiy, and Virgiliu Midrigan, "Inventories, Markups, and Real Rigidities in Menu Cost Models," *Review of Economic Studies*, 80 (2013), 249–276.
- Lins, Karl V., Henri Servaes, and Peter Tufano, "What Drives Corporate Liquidity? An International Survey of Cash Holdings and Lines of Credit," *Journal of Financial Economics*, 98 (2010), 160–176.
- Melcangi, Davide, "Firms' Precautionary Savings and Employment during a Credit Crisis," Working Paper, 2019.
- Mian, Atif, Kamalesh Rao, and Amir Sufi, "Household Balance Sheets, Consumption, and the Economic Slump," *Quarterly Journal of Economics*, 128 (2013), 1687–1726.
- Nakamura, Emi, and Jon Steinsson, "Lost in Transit: Product Replacement Bias and Pricing to Market," American Economic Review, 107 (2012), 3277–3316.
 - —, "Fiscal Stimulus in a Monetary Union: Evidence from US Regions," American Economic Review, 104 (2014), 753–792.
- Paravisini, Daniel, Veronica Rappoport, Philipp Schnabl, and Daniel Wolfenzon, "Dissecting the Effect of Credit Supply on Trade: Evidence from Matched Credit-Export Data," *Review of Economic Studies*, 82 (2015), 333–359.

- Peek, Joe, and Eric S. Rosengren, "The International Transmission of Financial Shocks: The Case of Japan," American Economic Review, 87 (1997), 495–505.
 - -, "Collateral Damage: Effects of the Japanese Bank Crisis on Real Activity in the United States," American Economic Review, 90 (2000), 30–45.
- Perez-Orive, Ander, "Credit Constraints, Firms' Precautionary Investment, and the Business Cycle," Journal of Monetary Economics, 78 (2016), 112-131.
- Phillips, Gordon, and Giorgo Sertsios, "How Do Firm Financial Conditions Affect Product Quality and Pricing?," *Management Science*, 59 (2013), 1764–1782. Rajan, Raghuram G., and Luigi Zingales, "Financial Dependence and Growth,"
- American Economic Review, 88 (1998), 559-586.
- Ramey, Valerie A., "Nonconvex Costs and the Behavior of Inventories," Journal of Political Economy, 99 (1991), 306-334.
- Ramey, Valerie A., and Kenneth D. West, "Inventories," Handbook of Macroeconomics, John B. Taylor and Michael Woodford, eds. (Amsterdam: Elsevier Science, 1999), 863-923.
- Santos, Joao, "Bank Corporate Loan Pricing Following the Subprime Crisis," Review of Financial Studies, 24 (2011), 1916-1943.
- Sufi, Amir, "Bank Lines of Credit in Corporate Finance: An Empirical Analysis," Review of Financial Studies, 22 (2009), 1057–1088.
- Wen, Yi, "Input and Output Inventory Dynamics," American Economic Journal: Macroeconomics, 3 (2011), 181–212. West, Kenneth D., "The Sources of Fluctuations in Aggregate Inventories and
- GNP," Quarterly Journal of Economics, 105 (1990), 939–971.
- Whited, Toni M., and Guojun Wu, "Financial Constraints Risk," Review of Financial Studies, 19 (2006), 531-559.